



Aichi-Nagoya 2026 Asian Para Games

Host City Agreement

October 3, 2023

Asian Paralympic Committee

P.O. Box 232111, 3/F Dubai Club for People of Determination, Al Nahda St., Al Qusais area, Dubai, United Arab Emirates

www.asianparalympic.org info@asianparalympic.org



Aichi-Nagoya 2026 Asian Para Games Host City Agreement

This Host City Agreement (hereinafter referred to as, this "**Agreement**") is, on October 3, 2023, made by and among:

- A. Asian Paralympic Committee (hereinafter referred to as, the "**APC**"), with its principal office at P.O. Box 232111, 3/F Dubai Club for People of Determination, Al Nahda St., Al Qusais area, Dubai, United Arab Emirates;
- B. Aichi Prefecture with its address at 3-1-2 Sannomaru, Naka-Ku, Nagoya, Aichi 460-8501 Japan and City of Nagoya with its address at 1-1 Sannomaru, 3-Chome, Naka-Ku, Nagoya, Aichi 460-8508 Japan (hereinafter collectively referred to as, the "City");
- C. Japanese Para-Sports Association with its principal office at 2-13-6 Nihonbashi-Kakigara-Cho, Chuou-Ku, Tokyo 103-0014 Japan (hereinafter referred to as, the "**JPSA**"); and
- D. Aichi-Nagoya Asian Games and Asian Para Games Organizing Committee with its principal office at 3-2-1 Sannomaru, Naka-Ku, Nagoya, Aichi 460-0001 Japan (hereinafter referred to as, the "**LOC**").

WHEREAS:

- 1. The APC is the regional organisation for the Asian Region (hereinafter referred to as, "**Region**") of the International Paralympic Committee (hereinafter referred to as, "**IPC**"), and is responsible for promoting the Paralympic Movement and Para sport events in the Region, in accordance with the APC Charter and the IPC Handbooks. The Asian Para Games is the exclusive property of the APC, and the APC is entitled to hold the Asian Para Games in the city of the Region selected and approved by the APC.
- 2. The Japanese Paralympic Committee (hereinafter referred to as, "**JPC**") is an internal organisation of the JPSA upon Article 43 of the articles of incorporation of JPSA. The JPC is the NPC for the Host Country and is recognised by the IPC as the sole representative of the Paralympic Movement in the Host Country.
- 3. The APC hereby approves that the LOC will organise and host the Asian Para Games 2026 in and around the City of the Host Country, the same City as the Asian Games 2026, in accordance with this Agreement.



- 4. The LOC agrees to organise the Asian Para Games 2026 in a professional manner for the benefit of Para athletes.
- 5. The LOC acknowledges the importance of the Asian Para Games 2026 as part of the APC's long-term objective of promoting and strengthening Para sports by exposing the Asian Para Games through the national and international media to a broad audience.
- 6. All Parties wish to promote the development of the values of the Paralympic Movement and of Para sports through the Asian Para Games 2026.

NOW, THEREFORE, IT IS AGREED AS FOLLOWS:

1 Term

This Agreement shall be effective on October 3, 2023 (hereinafter referred to as, "**Commencement Date**").

2 Appointment, Planning and Reporting

2.1 Appointment of LOC

- **2.1.1** The APC hereby appoints the LOC as the local organising committee for the Asian Para Games 2026.
- **2.1.2** The LOC will organise the Asian Para Games 2026 in a dignified manner as an independent Para sport event, and not in connection with any other international or national enterprise or event in the City, such as a fair or exhibition, with the exception of the Asian Games 2026.
- **2.1.3** The LOC shall not use the Asian Para Games 2026 to serve any purpose other than the interests of the Paralympic Movement and the development of the region (including, but not limited to, the development of the tourism and any other industries) in the Host Country.
- **2.1.4** The LOC acknowledges that the concerns for the accessibility and the environment are important considerations for the Asian Para Games 2026.
- **2.1.5** The LOC shall be directly involved in the management of the Asian Para Games 2026 through the employed staff, the directly managed consultants or any third parties outsourced to provide the services, for the Asian Para Games 2026.



2.1.6 The APC shall advise the LOC in relation to organising, planning and staging the Asian Para Games 2026. Whenever the LOC requests the APC to provide any advice upon the basis of the experiences in the previous Asian Para Games (including, but not limited to, advice on the operation and the financial plan of the Asian Para Games), the APC shall provide the LOC with such advice (including the provision of the materials) promptly. In addition, the LOC may, at its own authority, share the contents of the advice (including the provision of the materials) provided by the APC with the City and the JPSA.

2.2 Role of APC and JPC

- **2.2.1** As the representative of the Paralympic Movement in the Host Country, the JPC undertakes to work closely with the APC and the LOC to facilitate the organisation and staging of the Asian Para Games 2026.
- **2.2.2** As the host NPC, the JPC is responsible to ensure that a reasonable representation of athletes from the Host Country will participate in all the Paralympic sports among the Scheduled Sports as defined in Clause 4.2.1.
- **2.2.3** The APC is responsible for encouraging each NPC that a reasonable representation of athletes from the Region will participate in the Scheduled Sports as defined in Clause 4.2.1.

2.3 APC/IPC Regulations and Requirements

The LOC agrees to comply with the provisions of the APC Charter and the IPC Handbooks and the following APC/IPC regulations, requirements, and codes (hereinafter referred to as, "APC/IPC Regulations and Requirements"). In the event that it is difficult for the LOC to comply with some of APC/IPC Regulations and Requirements from the aspect of the budget, personnel and other factors of the Asian Para Games 2026, the LOC shall notify the APC thereof, and the APC and the LOC shall discuss the handling of such APC/IPC Regulations and Requirements in good faith.

- APC Anti-Doping Rules
- APC Athlete Classification Code
- IPC Athlete Nationality Code
- APC Code of Ethics
- IPC Medical Code



2.4 **Operational Requirements**

- **2.4.1** The LOC shall determine the following guidelines (hereinafter referred to as, "**Operational Requirements of Clause 2.4.1**") through separate discussion with the APC.
 - Asian Para Games 2026 Guidelines
 - Asian Para Games 2026 Torch Relay Guidelines
 - Best Practice Guide to Communications
- **2.4.2** The LOC agrees to respect the provisions outlined by the following documents (hereinafter referred to as, "**Operational Requirements of Clause 2.4.2**"). In the event that it is difficult for the LOC to respect some of the Operational Requirements of Clause 2.4.2 from the aspect of the budget, personnel and other factors of the Asian Para Games 2026, the LOC shall notify the APC thereof, and the APC and the LOC shall discuss the handling of such Operational Requirements of Clause 2.4.2 in good faith.
 - Paralympic Symbol Guidelines
 - IPC Regional Emblem Brand Book
 - IPC Guide to Para and IPC Terminology
 - IPC Technical Manual on Classification
 - IPC Data Exchange Process for Para Games Entries
 - IPC Annex Orthotic, Prosthetic and Wheelchair Repair Services at IPC Major Events
- **2.4.3** The LOC shall respect the policy of the IPC Accessibility Guide and make its efforts to secure the accessibility in the Asian Para Games 2026 through compliance with the Accessibility Standard as defined in Appendix B (hereinafter referred to as, "Accessibility Standard").

2.5 Change Management Process

2.5.1 All Parties recognise that, while the contents of the APC/IPC Regulations and Requirements and the Operational Requirements of Clause 2.4.2 represent the current position of the APC on such matters, such documents may evolve as a result of policy, technological and other changes (some of which may be beyond the control of All Parties). If some of the contents were amended, suspended or supplemented, the APC shall immediately notify the City, the JPSA and the LOC of such change in writing. The City, the JPSA and the LOC shall respect such change, considering the budget, personnel and other factors, so that the



Asian Para Games 2026 will be organised and presented in the best reasonably possible manner.

2.5.2 Notwithstanding Clause 2.5.1, if any of the City, the JPSA or the LOC considers that any amendments, suspensions or supplements made by the APC pursuant to Clause 2.5.1 result in adverse effects on its rights or obligations, the Party shall notify the other Parties (including the APC) thereof in writing within forty-five (45) days after the receipt of the APC's notice regarding the amendment, suspension or supplement made by the APC pursuant to Clause 2.5.1. The APC shall then negotiate with the City, the JPSA and the LOC to address such alleged adverse effects in a mutually satisfactory manner. If the negotiation is not settled, such amendment, suspension or supplement shall not be binding upon the City, the JPSA and the LOC.

2.6 Sport Rules and Guidelines

- **2.6.1** The Scheduled Sports as defined in Clause 4.2.1 are governed by the IF. Each IF shall adopt the IF Rules and use the IF Rules in the preparation for and during the Asian Para Games 2026.
- **2.6.2** The LOC agrees to comply with the IF Rules; provided, however, that in the event that it is difficult for the LOC to comply with some requirements of the IF Rules from the aspect of the budget, personnel and other factors of the Asian Para Games 2026, the LOC shall notify the IF and the APC of it, and the APC, the IF and the LOC shall discuss the handling of such requirements in good faith.

2.7 Sports Technical Memorandum of Understanding

The LOC will discuss with each IF a Sports Technical Memorandum of Understanding (hereinafter referred to as, "**STMOU**") for conclusion. Each STMOU shall outline the sports specific technical requirements for the Asian Para Games 2026, including, but not limited to, timelines, competition formats, competition equipment and officials.

2.8 Master Plan

Within six (6) months after the Commencement Date, the LOC shall submit to the APC for written approval a master plan of the Asian Para Games 2026 (hereinafter referred to as, "**Master Plan**"). Any



amendments to the Master Plan must be agreed in writing with the APC in advance.

2.9 Organisational Milestones

Within ninety (90) days after the Commencement Date, the APC shall provide the LOC with a master schedule of the organisational milestone dates for the specific tasks of the LOC outlined in the Operational Requirements of Clause 2.4.1 and the Operational Requirements of Clause 2.4.2.

2.10 Status Reports and Consultations

The LOC shall provide the APC with updates and details (in the form and volume discussed and determined between the APC and the LOC) regarding progress in accordance with the Master Plan (including, but not limited to, actual progress concerning the Asian Para Games 2026 Budget) every six (6) months, from twenty-four (24) months before the Opening Ceremony. Moreover, the LOC shall report the progress in writing or by email when the APC reasonably requests and the LOC determines the necessity of the report. In addition, the LOC shall establish consultation and communication with the designated representatives of the APC, the JPSA and the IF on a necessary basis.

2.11 Disclosures

If the APC requests the LOC to disclose the documents set forth in the following (a)(b)(c)(d)(e), the LOC will disclose the contents of such documents (e.g., transmitting the electronic data of the copy) to the APC in the method determined by the LOC's sole discretion.

- (a) Contracts with the directly managed consultants
- (b) Contracts with the third parties outsourced for providing services
- (c) Contracts with the LOC Marketing Agent as defined in Clause 10.1.2
- (d) Contracts with the LOC Partners as defined in Clause 10.1.3
- (e) Contracts with the LOC Licensee as defined in Clause 10.3.2

3 Grant of Commercial Rights

3.1 Commercial Rights and Commercial Revenues



- **3.1.1** Any and all commercial rights related to the Asian Para Games 2026 (hereinafter referred to as, "**Asian Para Games 2026 Commercial Rights**") shall consist of the Marketing Rights, the Media Rights and the Ticket Rights. The Marketing Rights shall consist of the Sponsorship Rights, the Licensing Rights and the Other Marketing Rights.
- **3.1.2** The APC grants the Asian Para Games 2026 Commercial Rights to the LOC; provided, however, that the APC reserves the rights of the Asian Para Games 2026 Performance Record as defined in Clause 5.4, the rights of the Asian Para Games 2026 Photograph as defined in Clause 5.5 and the rights of the GBO Broadcast Data set forth in Clause 11.3.1 among the Media Rights to the extent that these rights are stipulated in this Agreement.
- **3.1.3** Any and all revenues in any manner, including, but not limited to, cash and value-in-kind consideration (hereinafter referred to as, "**VIK**") derived from the Asian Para Games 2026 Commercial Rights (hereinafter referred to as, "**Asian Para Games 2026 Commercial Revenues**") shall consist of the Marketing Revenues (consisting of the Sponsorship Revenues, the Licensing Revenues and the Other Marketing Revenues), the Media Revenues and the Ticket Revenues.
- **3.1.4** The LOC shall obtain the following revenues among the Asian Para Games 2026 Commercial Revenues.
 - (a) the Sponsorship Revenues
 - (b) the Licensing Revenues (but excluding the NPC Distribution Money to the NPC upon Clause 10.3.3)
 - (c) the Other Marketing Revenues
 - (d) the Media Revenues
 - (e) the Ticket Revenues

3.2 Report of Revenues

Not later than twelve (12) months after the Closing Ceremony, the LOC must provide the APC with a breakdown of the revenues generated by the Asian Para Games 2026 in the manner determined by the LOC as part of the financial reporting requirements more specifically referred to in Clause 9.6.

4 The Asian Para Games 2026



4.1 **Period and Locations**

The Asian Para Games 2026 will consist of at least six (6) competition days and shall be held in and around the City. The Sport Classification Evaluation Period will be four (4) days or more before the competition days.

4.2 Sports, Events, Athletes, and NPC Team Officials

4.2.1 Sports

The sports to be scheduled in the Asian Para Games 2026 (hereinafter referred to as, "**Scheduled Sports**") are set forth in the Asian Para Games 2026 Sports Programme (as described in Appendix C). In the event that a Party intends to change the Scheduled Sports due to the difficult situations under which the Scheduled Sports are performed (e.g., lack of appropriate venues, insufficiency of budget or personnel, insufficiency of athletes' population), the Party shall offer the change with the reason to the other Parties and All Parties shall discuss the change. In the event that All Parties agree the change in writing, Appendix C shall be revised as such change is reflected. Provided, however, in the event of the change of the Scheduled Sports, the number of the Scheduled Sports on the Commencement Date. The specific events shall be separately discussed and determined among All Parties.

4.2.2 Test Events

In preparation for the Asian Para Games 2026, the LOC shall, in coordination with the JPSA, secure the opportunities for organising test events (including the tests events organised by any entities other than the LOC) for at least two (2) sports including the sports of Para athletics and Para swimming. The date and details for these test events must be agreed with the APC and the relevant IF.

4.2.3 Athletes and NPC Team Officials

The expected number of athletes who will participate in the Asia Para Games 2026 is between two thousand four hundred (2,400) and two thousand seven hundred (2,700). The expected number of NPC Team Officials shall be between one thousand two hundred (1,200) and one thousand three hundred (1,300). The number of the persons who are



with reduced mobility or daily wheelchair users among these athletes and NPC Team Officials shall be between eight hundred eighty (880) and nine hundred fifty (950). Provided, however, it is not possible on the Commencement Date to specify the number of the athletes and NPC Team Officials who will participate in the Asian Para Games 2026. The number of the foregoing athletes and NPC Team Officials who will participate in the Asian Para Games 2026 shall be finally determined through the discussion between the APC and the LOC eighteen (18) months prior to the Opening Ceremony. The acceptance and the terms of participation by athletes from non-APC member states shall be discussed and determined between the APC and the LOC.

4.2.4 Technical Delegates and Games Officials

The number of IF technical delegates and games officials (including, but not limited to judges, referees, classifiers, APC anti-doping officials, medical officials and TUE officials) shall be as mutually agreed between the APC, the LOC and the relevant IF, subject to the minimum requirements of the applicable IF rules and regulations, up to a maximum number of eight hundred (800). The APC and the IF shall make the best efforts to select appropriately qualified and experienced games officials from the Region.

4.3 Venues and Facilities

- **4.3.1** The Asian Para Games 2026 Venue will be separately discussed and determined by the APC and the LOC. The LOC shall make its efforts that the Asian Para Games 2026 Venues comply with the corresponding IF and APC requirements.
- **4.3.2** The LOC shall make its efforts that the Asian Para Games 2026 Venues meet the technical and minimal accessibility requirements established by the APC and each IF for their respective sport.
- **4.3.3** The LOC acknowledges the importance of accessible venues and infrastructure and will respect the ideas in the IPC Accessibility Guide.

5 Obligations of the LOC with regard to the Asian Para Games 2026

5.1 Accreditation



- **5.1.1** The Asian Para Games 2026 Identity and Accreditation Card (hereinafter referred to as, "**GIAC**") confers on the holder the right to attend the Asian Para Games 2026 and permits the holder access rights and other privileges, in order that he or she can perform specific functions connected with the Asian Para Games 2026. The LOC shall determine all the matters related to the GIAC, including, but not limited to, the issuance or cancellation of the GIAC, the classification of the GIAC and the contents of the privilege conferred by each GIAC, in coordination with the APC. "Accreditation" is the process of registering, producing, and distributing the GIAC.
- **5.1.2** The LOC shall plan and deliver the Asian Para Games 2026 accreditation services in accordance with the requirements stipulated in the Asian Para Games 2026 Accreditation Guide separately discussed and determined between the APC and the LOC and in accordance with the data processing agreement separately discussed and determined between the APC and the LOC. The LOC shall ensure that such services are compatible with the IPC Online Entry System, the IPC Sport Database Management System (hereinafter referred to as, "SDMS") and the IPC Accreditation System (collectively, hereinafter referred to as, "IPC Information Systems"). In addition, such services shall be suitably modified as discussed and determined between the APC and the LOC to ensure their accessibility.
- **5.1.3** The LOC shall bear all costs related to accreditation, including the costs of the production and distribution of the GIAC.
- **5.1.4** In the event that the APC pays any of the expenses that are set forth in Clause 5.1.3 above as the LOC's obligations upon the prior written approval of the LOC, it shall be entitled to invoice the LOC accordingly.

5.2 Entry Formalities

- **5.2.1** The LOC will approach the appropriate authorities about permitting access into the Host Country for the accredited persons on the basis of a passport [and /or equivalent document(s)] and the GIAC, without the need to obtain a visa within the scope of the relevant laws and the notifications from the government of the Host Country.
- **5.2.2** The LOC recognises that it will be necessary to arrange for the temporary entry, before, during and after the Asian Para Games 2026, of certain personnel into the Host Country and for the importation of equipment and supplies for purposes of the Asian Para Games 2026. The LOC will



approach the appropriate authorities about permitting such personnel to make entry into the Host Country. The LOC will approach the appropriate authorities about permitting that all such equipment and supplies, including medical supplies, will be allowed to enter the Host Country without any duties or taxes being payable. Provided, however, such equipment and supplies shall be either consumed for the Asian Para Games 2026 Period, disposed of (other than by sale) in the Host Country for the purpose set forth in this Agreement, or exported from the Host Country within a reasonable time after the Asian Para Games 2026.

5.3 Protocol Plans

The LOC will abide by all requirements in the Asian Para Games 2026 Protocol Guideline separately discussed and determined between the APC and the LOC and will submit the first version of the plans regarding the Asian Para Games 2026 Protocol to the APC (the plans regarding the protocol, such as the "Opening Ceremony", the "Closing Ceremony", the "Victory Ceremonies" and the "Team Welcome Ceremonies") no later than eighteen (18) months prior to the Opening Ceremony.

5.4 Rights in Ceremonies' Performances

The LOC shall execute the required documentation with the performers taking part in the Opening, Closing and Victory Ceremonies to ensure that the LOC is entitled to film and record their performances(hereinafter referred to as, "**Asian Para Games 2026 Performance Record**"), that the copyrights (including the translation rights, adaptation rights and other rights stipulated in Copyright Act Article 27 of the Host Country and the rights of the original author in connection with the exploitation of a derivative work stipulated in Copyright Act Article 28 of the Host Country) related to Asian Para Games 2026 Performance Record shall belong to the APC, that any rights other than the foregoing copyrights related to Asian Para Games 2026 Performance Record shall belong to the LOC, and that the APC is entitled to use the Asian Para Games 2026 Performance Record for the Asian Para Games 2026 and the Paralympic Movement without paying any rights fees, rental fees, or license fees.

5.5 Photographs



- 5.5.1 The copyrights (including the translation rights, adaptation rights and other rights stipulated in Copyright Act Article 27 of the Host Country and the rights of the original author in connection with the exploitation of a derivative work stipulated in Copyright Act Article 28 of the Host Country) related to any photographs taken by the LOC for the Asian Para Games 2026 Period and CD/DVD or external data storage of such photographs (collectively, hereinafter referred to as, "Asian Para Games 2026 Photograph") shall belong to the APC. Any rights other than the foregoing copyrights related to the Asian Para Games 2026 Photograph shall belong to the LOC. The LOC will ensure that a minimum of thirty (30) photographs per Scheduled Sports per day of the Asian Para Games 2026 (to be chosen by the APC) will be made available to the APC each day of the Asian Para Games 2026 with all rights cleared so that the APC will be entitled to make use of them for the purposes of promotion of the Asian Para Games and the Paralympic Movement.
- **5.5.2** The LOC will ensure that all photographs of the Asian Para Games 2026, as required by Clause 5.5.1, must be taken in accordance with the guidelines set forth in the IPC Guide to Reporting on Para athletes and must be a minimum of 300 dpi and 2250 x 1500 pixels. A CD/DVD or external data storage with the photos will be provided by the LOC to the APC.
- **5.5.3** The APC hereby agrees that the City will use the Asian Para Games 2026 Performance Record and the Asian Para Games 2026 Photograph, free of charge, over the world for the purpose as defined in Clause 8.5.2 (a) and (b) after the completion of the dissolution procedure of the LOC (for the avoidance of doubt, the procedure of Section 6 Dissolution of Chapter 3 General Incorporated Foundations and Chapter 4 Liquidation under Act on General Incorporated Associations and General Incorporated Foundations of the Host Country).

5.6 International Classification

The LOC will provide all the necessary facilities, venues and equipment for the work of the classification panels so that "Athlete Evaluation" (as defined in the APC Athlete Classification Code) can be carried out in accordance with the applicable IF classification rules and regulations for the Asian Para Games 2026 Period.

5.7 Insurance



- **5.7.1** The LOC will take out and maintain insurance for the Asian Para Games 2026. The LOC may determine all matters related to the insurance, including, but not limited to, the kinds of the insurance and the scope of the coverage, at the LOC's sole discretion. The LOC will provide the APC with the summary of the insurance in the manner determined by the LOC upon the request of the APC.
- **5.7.2** The APC shall request each NPC that the athletes participating in the Asian Para Games 2026 take out the insurance for the Asian Para Games 2026.

5.8 Travel and Transport

5.8.1 International Games Officials

The LOC will bear the expenses for return commercial economy air transportation incurred appropriately in the travel from each person's country of residence to the City on the occasion of the Asian Para Games 2026 for the following people (the total of (a)(b) below shall be up to a maximum number of eight hundred (800)):

- (a) All IF technical delegates; and
- (b) All international games officials.

The amount of such expenses actually borne by the LOC shall be determined through the discussion between the APC and the LOC.

5.8.2 Support to NPC

The LOC, in order to assist the NPCs, whose countries would have the difficulties of the participation in the Asian Para Games 2026, will provide free air ticket, free accommodation and/or discounted accommodation, to the extent the same have/has been provided as the VIK by the LOC Partners as defined in Clause 10.1.3 upon the LOC Sponsorship Agreement. The numbers of free air ticket, free accommodation and/or discounted accommodation will be determined by the LOC.

5.8.3 Transportation

The LOC shall provide safe and reliable ground transport to the holders of GIAC in accordance with the Asian Para Games 2026 Transportation Guideline and the Asian Para Games 2026 Accreditation Guideline separately discussed and determined between the APC and the LOC.

5.8.4 Transportation Plan



The LOC shall submit the first version of the transportation plan to the APC no later than twenty-four (24) months prior to the Opening Ceremony.

5.9 **Provision of Mobile Telephones**

The LOC will provide up to fifty (50) mobile telephones to the APC for their domestic use only (including data transmission) in connection with the Asian Para Games 2026. The charges for the mobile telephones thereof will be paid by the LOC.

5.10 Catering

5.10.1 Teams

The LOC will provide catering services consisting of suitable food and beverages for breakfast, lunch and dinner for all the athletes holding the GIAC and all the NPC Team Officials holding the GIAC. A suitable choice of meals, that incorporates specific nutritional and religious requirements, should be offered. Catering services must be provided in wheelchair accessible facilities at appropriate time taking into account the time of competitions.

5.10.2 APC Family

The LOC must make available catering services at the Asian Para Games 2026 Venue during Competition Sessions for the APC Family (as defined in Appendix D) in a designated lounge.

5.10.3 Media

The LOC must make available catering services at the Asian Para Games 2026 Venue and/or main media centre for the media. The LOC shall be entitled to charge reasonable prices for this service. Furthermore, the LOC shall provide the APC with the information regarding the prices for this service in the manner determined by the LOC upon the request of the APC.

5.10.4 Catering Plan

The LOC shall submit the first version of the catering plan to the APC no later than twelve (12) months prior to the Opening Ceremony.

5.11 Technology



- **5.11.1** The LOC is responsible for planning and implementing the technological services (hereinafter referred to as, "**Technological Services**") necessary for the successful preparation and operation of the Asian Para Games 2026. The Technological Services include the Games Management Systems ("GMS") (including online accreditation and entry systems), the On Venue Results ("OVR"), online results system, Information Diffusion Systems ("IDS") (including live web results), and Timing And Scoring Services ("T&S") in accordance with the Asian Para Games 2026 Results and Information Services ("APGRIS") requirements, and the IPC Data Exchange Process for Asian Para Games 2026 Entries requirements.
- **5.11.2** The LOC shall ensure that the Technological Services are compatible with the IPC Information Systems. In addition, the Technological Services shall be suitably modified as discussed and determined between the APC and the LOC to ensure their accessibility.
- **5.11.3** The LOC will appoint a consultant assisting in planning and implementing the Technological Services if the LOC determines that the consultant is necessary. The LOC may request the APC to provide the advice in appointing such consultant on a necessary basis.
- **5.11.4** In the event that the LOC appoints the technology providers to plan and implement the Technological Services, the LOC may request the APC to provide the advice on a necessary basis.
- **5.11.5** The LOC shall submit the first version of a technology plan/service implementation plan to the APC no later than twenty-four (24) months prior to the Opening Ceremony.

5.12 Look

- **5.12.1** The City and the LOC shall make a comprehensive "Look of the Asian Para Games 2026 Plan" for the Asian Para Games 2026, all the Asian Para Games 2026 Venue, and key locations throughout the City separately discussed and determined between the City and the LOC.
- **5.12.2** The Look of the Asian Para Games 2026 Plan must contain the Asian Para Games 2026 Marks as defined in Clause 8.3.1 and timelines for the delivery of them. Timelines must consider promotional events/activities. Furthermore, the Look of the Asian Para Games 2026 Plan must contain the plan of the Look Material.
- **5.12.3** All Look Material contained in the Look of the Asian Para Games 2026 Plan must follow the specifications provided in Paralympic Symbol



Guidelines, the Asian Para Games 2026 Look Guidelines separately discussed and determined between the APC and the LOC.

- **5.12.4** The LOC shall make its efforts to reflect a uniform image of the Asian Para Games 2026 and to position the Asian Para Games 2026 as a high-quality, high-standard event as befit the status of the Asian Para Games, considering the budget, personnel and other factors.
- **5.12.5** The LOC shall deliver the first version of the Look of the Asian Para Games 2026 Plan to the APC for approval no later than eighteen (18) months prior to the Opening Ceremony.

5.13 Sport Presentation

- **5.13.1** The LOC is responsible for implementing the sport presentation services necessary for the successful operation of the Asian Para Games 2026. "Sport Presentation" includes stadium announcements, music and entertainment, commentary (for the avoidance of doubt, the commentary is performed for not the broadcast but the spectators in the stadium), event videos, and other elements designed to create a unique and engaging spectator experience.
- **5.13.2** The LOC may request the APC to provide the advice on a necessary basis in selecting the providers for implementing the sport presentation services as defined in Clause 5.13.1.
- **5.13.3** The LOC shall submit the first version of the Sport Presentation Plan to the APC for approval no later than twelve (12) months prior to the Opening Ceremony.
- **5.13.4** Unless otherwise agreed between the APC and the LOC, the APC will nominate a consultant to assist the LOC to implement the requirements outlined in the Asian Para Games 2026 Sport Presentation Guidelines separately discussed and determined between the APC and the LOC. The APC will meet all the operational costs and expenses of this consultant, and the LOC will provide assistance and facilities reasonably requested by the consultant.

5.14 INFO System

The LOC shall provide an Asian Para Games 2026 "INFO System" that will be accessible to athletes, NPC Team Officials, and the APC Family as defined in Appendix D and media. The INFO System must include athlete biographical information. The LOC may request the APC to provide the



advice on a necessary basis in selecting the providers for providing the INFO System.

5.15 Security

Responsibility for all aspects of security is a matter to be dealt with by the appropriate authorities of the Host Country. The City and the LOC will approach the appropriate authorities about taking the appropriate and necessary security measures for the guarantee of the safe and peaceful celebration of the Asian Para Games 2026. The LOC shall submit the first version of the security plan to the APC twelve (12) months prior to the Opening Ceremony.

5.16 Each Person Appointed by the LOC

The LOC will take the necessary measures in coordination with the appropriate authorities so that the persons who are likely to hinder the safe and peaceful cerebration of the Asian Para Games 2026 are not included in each person (including all volunteers) appointed by the LOC to assist in hosting the Asian Para Games 2026.

5.17 Safety, Fire Protection and Medical Services

- **5.17.1** The City may determine the measures and level of the following items at its sole discretion, but shall do so at its own expense:
- **5.17.1.1** supporting the APC and the LOC in the following areas of safety and fire protection:
 - (a) outside of the Asian Para Games 2026 Venues (on any days when the Asian Para Games 2026 Venue is used in connection with the Asian Para Games 2026);
 - (b) at any Training Site (on such days when any training sessions in connection with the Asian Para Games 2026 take place);
 - (c) at the Asian Para Games 2026 Athletes' Village (for the Asian Para Games 2026 Period);
 - (d) at any other Asian Para Games 2026 Site (on such days when any Asian Para Games 2026-Related Events take place); and
 - (e) at the City's airports, main train stations as well as further public transportation sites, such as bus and metro stations as well as transportation vehicles, including airplanes, buses, metros and



trains, which are used in relation to the Asian Para Games 2026 for the Asian Para Games 2026 Period; and

- **5.17.1.2** adopting all general safety and security, fire protection and medical service measures as well as other protection measures in relation to the Asian Para Games 2026, including the provision of facilities, such as fences and other infrastructure and police protection for the key representatives of the APC Family as defined in Appendix D and VIP guests.
- **5.17.2** The City and the LOC shall be responsible for all aspects of medical/health services related to the Asian Para Games 2026, through the appropriate authorities in the City and the Host Country. Furthermore, the City and the LOC shall respect the policy as determined by the APC Medical & Sport Science Committee and be responsible for ensuring the implementation of all necessary and appropriate medical/health service measures, including repatriation. Medical services shall be provided free of charge to the holders of GIAC, to the extent that such medical services are necessary for the performance of the business in the Asian Para Games 2026 of each person, such as the participation in the Asian Para Games 2026 of the competitor, during his/her stay in the Host Country.
- **5.17.3** Any such general safety and security, fire protection and medical service measures shall be of the high quality and standard, and attended to with a high degree of priority.

5.18 Anti-Doping Programme

5.18.1 The LOC will include in the Master Plan an anti-doping programme, which is in accordance with the World Anti-Doping Code, the IPC Anti-Doping Code, the APC Anti-Doping Rules, the International Standards of World Anti-Doping Agency (hereinafter referred to as, "WADA") and the instructions of the APC Medical & Sport Science Committee Anti-Doping Sub-Committee. This programme will include the need to appoint an anti-doping organisation that will collect samples in full compliance with the APC Anti-Doping Rules and International Standard for Testing and Investigations (hereinafter referred to as, "ISTI"). This organisation must be recognised by WADA and use the WADA-accredited laboratory, located in the most convenient place in the Host Country and capable of analysing samples in accordance with the requirements of the APC Anti-Doping Rules, ISTI and the APC Medical & Sport Science Committee Anti-Doping Sub-Committee. The anti-doping organisation must be approved in writing by the APC in advance.



5.18.2 The LOC will support the implementation of an educational anti-doping programme by the APC, WADA and the Japan Anti-Doping Agency (JADA) in the Host Country.

5.19 Official Publications

The LOC shall submit to the APC for written approval a list of proposed Asian Para Games 2026 publications with referencing the list of the past Asian Para Games publications, no later than twenty-four (24) months prior to the Opening Ceremony.

5.20 Education and Training

5.20.1 The LOC will provide education and training to all volunteers and technical staff prior to the Asian Para Games 2026 (the detail and timelines of which will be submitted by the LOC to the APC for approval no later than eighteen (18) months prior to the Opening Ceremony).

5.20.2 Education Programme

All Parties will discuss the implementation of an education programme (targeting young people) to promote the inclusion of people with an impairment through Para sport. If All Parties agree to implement the education programme through the Asian Para Games 2026, the LOC shall use the best efforts to collaborate with the APC and the JPSA and use the IPC's global education programmes (including, but not limited to, I' m POSSIBLE).

5.20.3 Athlete Education Programme

All Parties will discuss the implementation of an athlete engagement area at the Asian Para Games 2026 (targeted at athletes competing in the Asian Para Games 2026). If All Parties agree to implement the athlete engagement area at the Asian Para Games 2026, the LOC shall use the best efforts to collaborate with the APC and the JPSA and use the Proud Paralympian Programme of the IPC to implement such area. The APC shall obtain the materials of the programme from the IPC, provide the LOC with the materials at no cost and assist the LOC in arranging for the lecturers.

5.21 Development Programmes

5.21.1 Community Programme



The City will plan and implement a Community Programme before and during the Asian Para Games 2026. The design of the Community Programme should focus on community participation in the lead-up to, and during, the Asian Para Games 2026. Furthermore, the Community Programme must contain a specific educational component involving school children that culminates in their attendance at the Asian Para Games 2026.

5.21.2 Para Sport Development Programme in the Region

All Parties will discuss the implementation of a Para Sport Development Programme titled: "Road to Aichi-Nagoya 2026." The purpose of the programme is designed to assist the NPC in the Region to develop programmes in relation to athlete participation, coaching, classification, officiating, event organisation and administration.

6 Accommodation

6.1 Asian Para Games 2026 Athletes' Village

- **6.1.1** The LOC shall provide accessible accommodation services and facilities (on a full room and board basis) in the Asian Para Games 2026 Athletes' Village. The Asian Para Games 2026 Athletes' Village will accommodate the athletes holding the GIAC between two thousand four hundred (2,400) and two thousand seven hundred (2,700), and the NPC Team Officials holding the GIAC between one thousand two hundred (1,200) and one thousand three hundred (1,300). The number of the persons who are with reduced mobility or daily wheelchair users among these athletes and NPC Team Officials will be between eight hundred eighty (880) and nine hundred fifty (950). The number of the foregoing athletes and NPC Team Officials will be finally determined through the discussion between the APC and the LOC eighteen (18) months prior to the Opening Ceremony.
- **6.1.2** In accordance with the Accessibility Standard, the LOC shall make the Asian Para Games 2026 Athletes' Village accessible by wheelchair users and consider the accessibility requirements of other disability groups participating in the Asian Para Games 2026.
- **6.1.3** The cost of boarding and lodging in the Asian Para Games 2026 Athlete's Village should not exceed fifty US Dollars (USD 50, including any and all taxes) per person per night.



- **6.1.4** The total quota for the athletes and the NPC Team Officials for each Team accommodated in the Asian Para Games 2026 Athletes' Village shall be established by the APC. Each quota for each facility of the Asian Para Games 2026 Athletes' Village shall be determined by the LOC.
- **6.1.5** If the APC and the LOC discuss and agree that the accessibility consultant (hereinafter referred to as, "**Accessibility Consultant**") to assist in satisfying the requirements separately discussed and determined between the APC and the LOC and the Accessibility Standard will be nominated, the LOC may select such Accessibility Consultant at its discretion. The LOC may request the APC to provide the advice on a necessary basis in selecting the Accessibility Consultant.
- **6.1.6** The LOC makes its efforts that the Asian Para Games 2026 Athletes' Village will be located in the same facilities used by the Team of the National Olympic Committees during the Asian Games 2026. Provided, however, the LOC will determine the facilities as the Asian Para Games 2026 Athletes' Village, considering the final location of the Asian Games 2026 Venue and the Asian Para Games 2026 Venue, and the reservations and any other circumstances of the hotels designated as the Asian Para Games 2026 Athletes' Village and any other factors.
- **6.1.7** The LOC shall make the Asian Para Games 2026 Athletes' Village (and all related services) available for the Asian Para Games 2026, no later than seven (7) days prior to the Opening Ceremony and until two (2) days following the Closing Ceremony (If the hotel rooms are provided for the Asian Para Games 2026 Athletes' Village, the available time shall follow the check-in time and check-out time determined by the hotels); provided, however, that the period of this service may be shortened through discussion between the APC and the LOC, based on the programmes of the Asian Para Games 2026.
- **6.1.8** The scope and level of services at the Asian Para Games 2026 Athletes' Village must comply with the requirements separately discussed and determined between the APC and the LOC and the Accessibility Standard.
- **6.1.9** Unless the APC and the LOC agree otherwise, the LOC makes its efforts, considering the circumstances of the facilities designated as the Asian Para Games 2026 Athletes' Village, that no other persons or groups will be accommodated in the specific area of such facilities.

6.2 Accommodation other than Asian Para Games 2026 Athletes' Village



The LOC shall make its efforts to ensure that the accommodation other than the Asian Para Games 2026 Athletes' Village will consider the accessibility requirements for each disability group of all APC Family as defined in Appendix D, games officials and media personnel.

6.3 Asian Para Games 2026 Accommodation Plans

The LOC shall submit the first version of an accommodation plan to the APC for written approval no later than eighteen (18) months prior to the Opening Ceremony. The LOC shall obtain the APC's approval for the selected hotels before entering into any contracts with them.

6.4 APC Family

- **6.4.1** The LOC shall provide the rooms, on a full board basis, for the members of the APC Family (as defined in Appendix D) in the Asian Para Games 2026 official hotel (hereinafter referred to as, "**APC Family Hotel**") for the Asian Para Games 2026 Period. The conditions of the APC Family Hotel shall be discussed and determined between the APC and the LOC. Provided, however, on the Commencement Date, it is duly acknowledged that it is impossible to specify the number of the members of the APC Family who will stay at the rooms and the period of stay at the rooms of these persons. Hence, the final report of the above number and the period of stay at the rooms shall be provided by the APC to the LOC no later than twelve (12) months prior to the Opening Ceremony.
- **6.4.2** The maximum number of room-nights on a full board basis, free of charge, to be provided by the LOC for the following members of the APC Family will be eight hundred (800). Provided, however, if the members of the APC Family stay at the APC Family Hotel, in excess of eight hundred (800) in the maximum number of room-nights, the LOC makes its efforts that the excess for the accommodation will be provided at the reasonable price separately discussed and determined between the APC and the LOC.
 - (a) APC Executive Board Members
 - (b) APC Standing Committee Chairpersons
 - (c) APC Standing Committee Members
 - (d) APC Staff
 - (e) Guides/Assistants/Translators for APC Executive Board and Standing Committee Members
 - (f) APC Guests



- **6.4.3** The LOC will provide the rooms of the APC Family Hotel for the members of the APC Family not set forth in the items from (a) to (f) of Clause 6.4.2 for a fee.
- **6.4.4** The APC may require the LOC to provide accommodation for the APC staff set forth in Clause 6.4.2 (d) in the same facilities as the Asian Para Games 2026 Athletes' Village in place of the APC Family Hotel.

6.5 Games Officials

- **6.5.1** The LOC shall provide the hotel rooms, on a full board basis, (hereinafter referred to as, "**Games Officials Hotel**"), free of charge, for a maximum of eight hundred (800) people, which include the IF technical delegates and international and national games officials. The LOC shall secure the Games Officials Hotel in accordance with the conditions set forth in the following (a)(b)(c). The scope of the persons who will be provided the Games Officials Hotel and other conditions of the Games Officials Hotel will be discussed and determined between the APC and the LOC.
 - (a) IF technical delegates: one (1) person per room
 - (b) International games officials: one (1) person per room
 - (c) National games officials: one (1) person or two (2) persons per room
- **6.5.2** The LOC may provide the accommodations on a full room and board basis as a part of the Games Officials Hotel as defined in Clause 6.5.1 for the IF technical delegates and games officials for the Asian Para Games 2026 Period in the same facilities as the Asian Para Games 2026 Athletes' Village. The accommodations must be in a housing section separate from that of the housing section for the athletes holding the GIAC and the NPC Team Officials holding the GIAC.

6.6 Media

The LOC shall facilitate access to suitable hotel accommodation with reasonable room rates for the media personnel holding the GIAC. It will be the responsibility of the media personnel to book and pay for their own accommodation. Unless agreed otherwise in writing with the APC, this group must have different accommodation than the athletes and the NPC Team Officials.



7 Meetings and Visits

7.1 Asian Para Games 2026 Coordination Meetings

- **7.1.1** The LOC shall arrange for a maximum of two (2) Asian Para Games 2026 Co-ordination Meeting (hereinafter referred to as, "**CoComs**") per year (a maximum of six (6) in total), to be held in the City for the purpose that the APC supervises and supports the LOC. If the APC holds the CoComs in the City, the APC shall notify the LOC of it no later than three (3) months prior to the meeting day. The LOC shall cover the reasonable costs associated with these meetings including hotels on a full board basis, economy air travel, ground transport and the facilities and technology for a maximum of eight (8) representatives of the APC for a maximum of three (3) nights per meeting. The scope and amount of such expense actually borne by the LOC shall be discussed and determined between the APC and the LOC in advance whenever the CoComs is held.
- **7.1.2** Notwithstanding Clause 7.1.1, the APC and the LOC may hold the CoComs in the manner of Web conference through discussion, considering various circumstances. In the case of the manner of a web conference, the LOC shall bear only reasonable expense (including arranging for the translator and the conference room) incurred in the Host Country for holding the CoComs. The scope and amount of such expense actually borne by the LOC shall be discussed and determined between the APC and the LOC.

7.2 Technical Site Visits

The LOC shall organise a maximum of two (2) technical site visits for each IF of the Scheduled Sports in and around the City; provided, however, that if each IF intends additional technical visit and such visit is required upon the discussion and the determination between the APC and the LOC, the LOC will organise such visit on a necessary basis. The LOC shall bear the reasonable costs associated with these visits including hotels on a full board basis, economy air travel, ground transport and facilities and technology for up to two (2) representatives of each IF for a maximum of three (3) nights per visit. The scope and amount of such expense actually borne by the LOC shall be discussed and determined between the APC and the LOC.

7.3 Facilities for APC Meetings for the Asian Para Games 2026 Period



The LOC shall provide suitable meeting rooms, audio-visual and technical equipment and transportation services in the Host Country necessary to conduct the various APC technical meetings for the Asian Para Games 2026 Period in accordance with the Meetings Organisation Guide separately discussed and determined between the APC and the LOC.

7.4 APC Executive Board Meeting

- **7.4.1** The LOC shall assist in organising one (1) APC Executive Board Meeting. This will take place in the City two (2) days prior to the day of the Opening Ceremony (or such other date as reasonably agreed between the APC and the LOC). The LOC shall also assist in organising, in consultation with the APC, all the meetings associated with the APC Executive Board Meeting on the dates to be agreed between the APC and the LOC. The expense and cost related to convening and implementing such meetings shall be shared by the APC and the LOC through discussion, based on the principle that the LOC shall be responsible for the meals, accommodations, conference rooms and transportation in the Host Country and the APC shall be responsible for all the international air transportation.
- **7.4.2** The LOC shall ensure that the accessible facilities with the technology separately discussed and determined between the APC and the LOC are available for the meetings set forth in Clause 7.4.1.

7.5 APC Athletes Committee Election

The APC Athletes Committee elections will take place in the manner separately discussed and determined between the APC and the LOC for the Asian Para Games 2026 Period. The APC shall organise the APC athlete elections, and the LOC shall provide the assistance (including the provision of the necessary technical equipment) reasonably requested by the APC in accordance with the Meetings Organisation Guide separately discussed and determined between the APC and the LOC.

8 Intellectual Property Rights

8.1 Categories of Intellectual Property Rights

8.1.1 The APC shall have the ownership of all the proprietary rights, including, but not limited to, trademark, copyright, design right and any other intellectual property rights in the following items (collectively, hereinafter



referred to as, "APC-IP").

- (a) Name of Asian Para Games
- (b) APC Trademarks and Logos
- (c) APC Flags
- (d) APC Sport Graphics
- (e) Emblems, mascots and other proprietary rights of the previous Asian Para Games which are in the sole ownership of the APC
- (f) All rights of whatever kind or nature in respect of other indicia used in the context of the APC (e.g., APC vision and mission) (but excluding the LOC-IP of Clause 8.1.2)
- (g) APC's rights related to the Asian Para Game 2026 Performance Record as defined in Clause 5.4, APC's rights related to the Asian Para Game 2026 Photograph as defined in Clause 5.5, and APC's rights related to the GBO Broadcast Data as defined in Clause 11.3.1
- (h) All derivative works using the works specified in items of (b), (c), (d), (e), (f) and /or (g) above
- **8.1.2** The LOC shall have the ownership of all the proprietary rights, including, but not limited to trademark, copyright, design right and any other intellectual property rights in the following items as to the Asian Para Games 2026 (collectively, hereinafter referred to as, "LOC-IP").
 - (a) Name of Asian Para Games 2026
 - (b) Asian Para Games Marks 2026 of Clause 8.3.1
 - (c) Asian Para Games 2026 Required Designs of Clause 8.3.2
 - (d) Asian Para Games 2026 Optional Designs of Clause 8.3.4
 - (e) Asian Para Games 2026 Official Programmes and other publications
 - (f) Asian Para Games 2026 Official Websites
 - (g) LOC's rights related to the Asian Para Game 2026 Performance Record as defined in Clause 5.4, LOC's rights related to the Asian Para Game 2026 Photograph as defined in Clause 5.5, and LOC's rights related to the GBO Broadcast Data as defined in Clause 11.3.1
 - (h) All derivative works using the works specified in items of (b), (c), (d), (e) and /or (f) above

8.2 Representation, Warranty and Protection Measures of Intellectual Property Rights

8.2.1 The APC hereby represents and warrants that:



- (a) the APC is the sole owner and holder of the APC-IP;
- (b) the APC-IP has not infringed upon any rights of any third parties; and
- (c) there have been no claims or disputes with respect to the APC-IP between the APC and any third parties.
- **8.2.2** The APC shall take the appropriate legal protection measures for the protection (including trademark registration) of the APC-IP in the members countries (including the Host Country) of the APC at its own costs and expenses.
- **8.2.3** The LOC hereby represents and warrants that:
 - (a) the LOC shall be the sole owner and holder of the LOC-IP (but excluding a part of some rights which the third party holds upon the agreement with the third party, or the laws and regulations) for the period beginning on the day when the rights in relation to such LOC-IP are obtained by the LOC and ending on the day when such LOC-IP is transferred to the APC (including the Asian Para Games 2026 Period); and
 - (b) the LOC-IP has not infringed upon any rights of any third parties.
- **8.2.4** The LOC shall take the appropriate legal protection measures for the protection (including trademark registration) of the LOC-IP at its own costs and expenses in accordance with Clause 8.7.1.

8.3 Intellectual Property Rights of Asian Para Games 2026

- 8.3.1 The LOC will, in consultation with and with the written approval of the APC, no later than eighteen (18) months prior to the Opening Ceremony, design the following indicia (collectively, hereinafter referred to as, "Asian Para Games 2026 Marks") which will incorporate the words "Aichi-Nagoya 2026 Asian Para Games".
 - (a) Asian Para Games 2026 Emblems
 - (b) Asian Para Games 2026 Mascots
 - (c) Asian Para Games 2026 Logos
 - (d) Other indicia related to the Asian Para Games 2026 discussed and determined between the APC and the LOC

The LOC will ensure that it will use the approved Asian Para Games 2026 Marks at all times in a logical and consistent manner when referring to and in connection with the Asian Para Games 2026, and in the Look elements.





- **8.3.2** The LOC shall design the following graphic, visual, artistic and intellectual works with respect to the Asian Para Games 2026 (hereinafter referred to as, "Asian Para Games 2026 Required Designs").
 - (a) Asian Para Games 2026 Victory Medals and any Moulds relating thereto
 - (b) Asian Para Games 2026 Victory and Athletes Diplomas
 - (c) Asian Para Games 2026 Accreditation Cards and Lanyards
 - (d) Asian Para Games 2026 Tickets
- **8.3.3** The number of strikes on all Asian Para Games 2026 Victory Medals shall be subject to the prior written approval of the APC. The LOC shall provide the APC with a certificate from the person responsible for producing such medals that the permitted number of strikes has been respected.
- 8.3.4 In addition, the LOC may create other Asian Para Games 2026 Designs (including, but not limited to, torches, pictograms, audio visual and musical works.) (hereinafter referred to as, "Asian Games 2026 Optional Designs").

8.4 Transfer of LOC-IP to APC

The LOC hereby agrees to transfer, free of charge, the LOC-IP (but excluding a part of some rights which the third party holds upon the agreement with the third party, or the laws and regulations) to the APC in the dissolution procedure of the LOC (for the avoidance of doubt, the procedure of Section 6 Dissolution of Chapter 3 General Incorporated Foundations and Chapter 4 Liquidation under Act on General Incorporated Associations and General Incorporated Foundations of the Host Country) after the completion of the Asian Para Games 2026. However, the APC shall bear the costs and expenses for the registration (e.g., change of the trademark holder in registration) of the intellectual property rights in the LOC-IP for such transfer, and the LOC shall assist the APC in taking such registration procedure. In the event that the trademark holder in registration of the Host Country is changed from the LOC to the APC, the APC shall appoint a trademark administrator (e.g., patent attorney ("Benrishi") in the Host Country), a representative acting for the APC in handling matters (including the renewal registration procedure upon Clause 8.7.2) related to such trademark and notify the City of the name and address of the trademark administrator in writing.

8.5 Use of APC-IP



- 8.5.1 The APC hereby grants to the LOC a non-exclusive license and right to use and work the APC-IP, with the right to grant a sub-license, solely for the purpose of preparation, management, promotion and operation of the Asian Para Games 2026, and any other activities in connection with the Asian Para Games 2026, the Asian Para Games 2026-Related Events and the Host City Events over the world, free of charge, for the period beginning on the Commencement Date and ending on the completion of the dissolution procedure of the LOC (for the avoidance of doubt, on the completion of the procedure of Chapter 4 Liquidation under Act on General Incorporated Associations General and Incorporated Foundations of the Host Country) after the completion of the Asian Para Games 2026. Furthermore, the APC hereby grants to the City a nonexclusive license and right to use and work the APC-IP for the purpose of preparation, management, promotion and operation of the Asian Para Games 2026, and any other activities in connection with the Asian Para Games 2026, the Asian Para Games 2026-Related Events and the Host City Events over the world for such period, free of charge.
- **8.5.2** Upon the completion of the dissolution procedure of the LOC (for the avoidance of doubt, upon the completion of the procedure of Chapter 4 Liquidation under Act on General Incorporated Associations and General Incorporated Foundations of the Host Country) after the completion of the Asian Para Games 2026, the APC hereby agrees that the City shall have, free of charge, a non-exclusive license and right to use and work the APC-IP as long as such APC-IP is legally valid and effective over the world in the following cases:
 - (a) the exhibitions and commemorative exhibitions in connection with the Asian Para Games 2026; and
 - (b) the activities for the promotion of the City.
- **8.5.3** In the event that the APC allows any third party to use and/or work the APC-IP in any manner (including, but not limited to, the grant of any license and right of the APC-IP as to the Asian Youth Para Games) in the Host Country, the APC shall report such third party, the scope of such license and right, and any other related information to the LOC.

8.6 Use of LOC-IP

8.6.1 The LOC, hereby grants to the APC and the City a non-exclusive license to use the LOC-IP, free of charge, for the period beginning on the day when the LOC obtains rights in relation to the LOC-IP and ending on the completion day of the dissolution procedure of the LOC (for the avoidance



of doubt, on the completion day of the procedure of Chapter 4 Liquidation under Act on General Incorporated Associations and General Incorporated Foundations of the Host Country) after the completion of the Asian Para Games 2026 upon the prior approval of the LOC in the following cases:

- (a) the activities for the advertisement and diffusion of the Asian Para Games 2026; and
- (b) the promotion activities of the APC and/or the City.
- 8.6.2 At the time of the transfer of the LOC-IP by the LOC to the APC set forth in Clause 8.4, the APC hereby agrees to grant to the City, free of charge, an exclusive license and right to use and work the LOC-IP (including the manufacture and sale of merchandise with the use of the Asian Para Games 2026 Emblems and the Asian Para Games 2026 Mascots and any other LOC-IP of the Asian Para Games 2026) with the right to grant a sub-license, over the world as long as such LOC-IP is legally valid and effective. For the avoidance of doubt, in relation to such merchandise, the City may use the APC's trademarks and logos, free of charge. The APC agrees to grant to the City the exclusive license and right (e.g., Senyo-Jisshiken in Design Right; Senyo-Shiyoken in Trademark Right) of the trademarks, design rights and any other intellectual property rights in the Host Country with respect to the LOC-IP. The City may register such exclusive license and right (Senyo-Jisshiken; Senyo-Shiyoken) at its own costs and expenses in the Host Country, and the APC shall assist the City in taking such registration procedure.
- **8.6.3** The City and the LOC shall comply with the requirements of IPC Guide to Para and IPC Terminology, Paralympic Symbol Guidelines and IPC Regional Emblem Brand Book in using the APC-IP and the LOC-IP.

8.7 Protection Measures of LOC-IP

- **8.7.1** The LOC shall take the appropriate measures (including the trademark registration) to protect the LOC-IP at its own costs and expenses as follows:
 - (a) the LOC shall register the trademark with respect to the Name of Asian Para Games 2026 within twelve (12) months after the Commencement Date, and trademarks with respect to the Asian Para Games 2026 Emblems and the Asian Para Games 2026 Mascots promptly after the determination thereof, in accordance with the Trademark Act in the Host Country;



- (b) the LOC shall take the appropriate legal protection of trademark registration or other protection measures equivalent to such registration with respect to the Name of Asian Para Games 2026, the Asian Para Games 2026 Emblems and the Asian Para Games 2026 Mascots in the regions outside the Host Country determined by the LOC; and
- (c) the LOC shall take the appropriate measures to protect the LOC-IP other than the trademarks set forth in Paragraphs (a) and (b) above within the necessary scope of the protection for such LOC-IP at its sole discretion.
- **8.7.2** In the event that the City determines that it is necessary to register the renewal procedure of LOC-IP in accordance with Clause 8.6.2, the City may request the APC to undertake such renewal registration procedure and the APC shall cooperate with it. The costs and expenses for such renewal registration procedure shall be borne by the City.

8.8 Infringement of APC-IP

In the event that the City, the JPSA and/or the LOC become(s) aware of any infringement of APC-IP by any third party, such as an unauthorized use of the name of "the Asian Para Games" or APC Logos, or any Ambush Marketing, the City, the JPSA and/or the LOC shall notify the APC, without unreasonable delay, of such infringement, of the following details in particular:

- (a) the identity of the infringing party (if apparent);
- (b) the nature of the activities;
- (c) where the activities have taken place;
- (d) the date(s) on which the activities were discovered; and
- (e) any other information requested by the APC (if apparent).

The APC shall, at its own costs and expenses, take the necessary countermeasures against the infringement upon the APC-IP by any third party, and the LOC shall assist the APC in the APC's taking such countermeasures.

8.9 Infringement of LOC-IP

8.9.1 For the period when the LOC owns the LOC-IP upon Clause 8.1.2, in the event that the APC, the City and/or the JPSA become(s) aware of any infringement of the LOC-IP by any third party (including through the report to the APC by the NPC), such as an unauthorized use of the Name



of Asian Para Games 2026, the Asian Para Games 2026 Emblems or the Asian Para Games 2026 Mascots, or any Ambush Marketing, the APC, the City and/or the JPSA shall notify the LOC, without unreasonable delay, of such infringement, of the following details in particular:

- (a) the identity of the infringing party (if apparent);
- (b) the nature of the activities;
- (c) where the activities have taken place;
- (d) the date(s) on which the activities were discovered; and
- (e) any other information requested by the LOC (if apparent).

The LOC shall, at its own costs and expenses, take the necessary countermeasures against the infringement upon the LOC-IP by any third party, and the APC shall assist the LOC in the LOC's taking such countermeasures.

- **8.9.2** For the period when the City owns an exclusive license and right of the LOC-IP upon Clause 8.6.2, in the event that the APC and/or the JPSA become(s) aware of any infringement of the LOC-IP by any third party (including through the report to the APC by the NPC), such as an unauthorized use of the Name of Asian Para Games 2026, the Asian Para Games 2026 Emblems or the Asian Para Games 2026 Mascots, or any Ambush Marketing, the APC and/or the JPSA shall notify the APC and the City, without unreasonable delay, of such infringement, in particular of the following details:
 - (a) the identity of the infringing party (if apparent);
 - (b) the nature of the activities;
 - (c) where the activities have taken place;
 - (d) the date(s) on which the activities were discovered; and
 - (e) any other information requested by the APC and the City (if apparent).

The City shall, at its own costs and expenses, take the necessary countermeasures against the infringement upon the LOC-IP by any third party (e.g., injunction procedure against the infringement upon the registered Senyo-Shiyoken of the trademark right) under the applicable law in the Host Country as long as the City owns an exclusive license and right of the LOC-IP upon Clause 8.6.2. The APC shall assist the City in the City's taking such countermeasures. However, in the event that it is difficult for the City to take the necessary countermeasures against such infringement in the Host Country, the APC shall take the necessary countermeasures upon the request by the City. The APC and the City shall take the necessary countermeasures against the infringement upon



the LOC-IP by any third party outside the Host Country through the discussion between the APC and the City as long as the City owns an exclusive license and right of the LOC-IP upe 8.6.2.

9 Financial Arrangements

9.1 Games Holding Fee

- **9.1.1** The LOC agrees to pay to the APC a total amount of three million US Dollars (USD 3,000,000) as a Games Holding Fee (hereinafter referred to as, "**Games Holding Fee**") in consideration of holding the Asian Para Games 2026. The Games Holding Fee shall consist of the fee of the following (a)(b)(c), and the details of each fee shall be defined in Appendix E. For the avoidance of doubt, unless otherwise specified in this Agreement, any and all rights granted by the APC to the LOC upon this Agreement shall be released on the Commencement Date. The handling of the tax imposed upon the payment of this three million US Dollars (USD 3,000,000) as the Games Holding Fee shall be subject to Clause 9.4, and the payment will be made by the LOC that the APC receives into its bank account the total sum of three million US Dollars (USD 3,000,000).
 - (a) Games Sanctioning Fee
 - (b) Fee for Administrating, Planning and Advising
 - (c) Fee for Granting the Asian Para Games 2026 Commercial Rights
- **9.1.2** The LOC shall pay the Games Holding Fee in accordance with the following payment schedule. The details of each payment shall be defined in Appendix E; provided, however, the handling of the tax imposed upon the payment of this Games Holding Fee shall be subject to Clause 9.4.
 - (a) One million US Dollars (USD 1,000,000) on or before December 31, 2023 as a part of the Games Sanctioning Fee set forth in Appendix E.
 - (b) One million US Dollars (USD 1,000,000) on or before June 30, 2025 as a part of the Games Sanctioning Fee set forth in Appendix E.
 - (c) One million US Dollars (USD 1,000,000) on or before August 31, 2026 as a part of the Games Sanctioning Fee (USD 300,000), all of the Fee for Administrating, Planning and Advising (USD 300,000), and all of the Fee for Granting the Asian Para Games 2026 Commercial Rights (USD 400,000) set forth in Appendix E.



9.2 Per Diems

The LOC will provide each IF technical official and games official, up to a maximum of eight hundred (800), with a per diem for the entire period of their stay in the City or the designated place in the Host Country for the Asian Para Games 2026, including travel days to and from the City or such place. The amount and scope which the LOC actually bears for the per diems will be discussed and determined by the APC and the LOC.

9.3 Capitation Fee

- **9.3.1** The APC may charge a capitation fee of fifty US Dollars (USD 50) (hereinafter referred to as, "**Capitation Fee**") to each category member of each Team holding the GIAC.
- **9.3.2** The LOC has no right to charge any entry or participation fees to any of the members of the participating Team.

9.4 Taxes

The LOC shall bear any withholding taxes or other taxes, customs duties, or any other type of governmental or other charge in the Host Country imposed on the payments made by the LOC to the APC in or related to this Agreement.

9.5 Budget

- 9.5.1 The LOC will produce and provide the APC with a summarized Asian Para Games 2026 Budget (in US dollars) (hereinafter referred to as, "Financial Plan") within nine (9) months after the Commencement Date. This must include explanatory notes and rationale on the expected revenue and expenses.
- **9.5.2** All Parties shall make the efforts to operate the Asian Para Games 2026 reasonably within the scope of the Financial Plan.
- **9.5.3** If the LOC recognizes that any event of the following (a)(b)(c)(d) will give adverse effects (e.g., increase of costs, decrease of revenues) upon the Financial Plan and notifies the other Parties thereof, All Parties shall discuss the resolution (including, but not limited to, the cancellation, relocation, advancement, postponement or interruption of one or more Asian Para Games 2026 Events or Asian Para Games 2026-Related



Events, and/or the reduction of various service levels or the cancellation of various services) for it:

- (a) Increase of prices, or increase of construction material and transportation costs etc. in the Host Country;
- (b) Depreciation of Japanese Yen against US Dollars due to exchange rate fluctuation;
- (c) One or more NPC(s) joining in the APC, or one or more IF and/or AF has (have) expressed its/their intention of non-participation in the Asian Para Games 2026; or
- (d) One or more NPC(s) joining in the APC, or one or more IF and/or AF has (have) been prohibited from participation in the Asian Para Games 2026 due to penalty or punishment.
- **9.5.4** If the LOC has any material revisions to the Financial Plan (revisions exceeding a plus/minus 10% effect on the projected profit/loss), the LOC shall provide the APC with the revised Asian Para Games 2026 Budget.
- **9.5.5** The LOC must ensure that it arranges its Asian Para Games 2026 Budget in a way that clearly identifies the separate income and expenditure relating to the Asian Games 2026 and the Asian Para Games 2026.

9.6 Financial Reporting

The LOC agrees to the following financial reporting requirements (a)(b):

- (a) Submitting annual finance reports (in the form discussed and determined by the APC and the LOC) to the APC; and
- (b) Submitting no later than twelve (12) months after the Closing Ceremony, a final audited account of the Asian Para Games 2026 to the APC indicating all income and expenditure (with adequate descriptions of all items).

10 Marketing

10.1 LOC Asian Para Games 2026 Marketing Programme

10.1.1 The LOC is entitled to, at its sole discretion, develop and work out the marketing programme with respect to the Asian Para Games 2026 Commercial Rights (hereinafter referred to as, "LOC Asian Para Games 2026 Marketing Programme") on a necessary basis. For the avoidance of doubt, the territory of the LOC Asian Para Games 2026 Marketing Programme shall be inside and outside the Host Country. Upon the request of the LOC to the APC with respect to developing and working



out the LOC Asian Para Games 2026 Marketing Programme, the APC shall provide the LOC with the necessary information and any other assistance.

- **10.1.2** The LOC may, at its sole discretion, nominate and appoint a marketing agent (hereinafter referred to as, "LOC Marketing Agent") as its sales and marketing agent to develop, work out and perform the LOC Asian Para Games 2026 Marketing Programme, and determine the contents of the contract, including, but not limited to, the scope of the authority entrusted to the LOC Marketing Agent and the commission of the LOC Marketing Agent, with the LOC Marketing Agent.
- **10.1.3** The LOC may, at its sole discretion, enter into the LOC Sponsorship Agreement with the third parties (hereinafter referred to as, "LOC **Partners**"). If the LOC enters into the LOC Sponsorship Agreement with the new LOC Partners, the LOC shall promptly give its notice thereof to the APC. For the avoidance of doubt, the LOC may grant the use and work of the APC-IP to the LOC Partners in the LOC Sponsorship Agreement as the sub-license rights of the APC-IP without any payment to the APC upon Clause 8.5.1.
- **10.1.4** The LOC Partners may do Sponsorship Activation inside and outside the Host Country.
- **10.1.5** If the APC is aware of the parties (including the APC Partners) who have an interest in becoming the LOC Partners (or otherwise becoming associated with it), the APC shall introduce the LOC such parties.
- **10.1.6** If the LOC is aware that the LOC Partners offer to sponsor the APC beyond its commitment with the Asian Para Games 2026, the LOC shall inform the APC of it.
- **10.1.7** The APC shall support the LOC to be able to enter into the LOC Sponsorship Agreement with the IPC Worldwide Paralympic Partners and the APC Partners if the LOC desires to do so.

10.2 Asian Para Games 2026 Partner Categories

- **10.2.1** The LOC shall set up the Asian Para Games 2026 Partner Categories at its sole discretion and notify the APC of the contents thereof.
- **10.2.2** The APC hereby represents that the APC Partners on the Commencement Date (hereinafter referred to as, "**Commencement Date APC Partners**") are defined in Appendix F. If the Commencement Date APC Partners desire to enter into the LOC Sponsorship Agreement, the LOC shall discuss with the Commencement Date APC Partners for entering



into the LOC Sponsorship Agreement on a priority basis. With respect to any Asian Para Games 2026 Partner Categories other than the category to which the Commencement Date APC Partners belong, the LOC may discuss with any third parties other than the Commencement Date APC Partners and enter into the LOC Sponsorship Agreement with such third parties. With respect to the category to which the Commencement Date APC Partners belong, if such Commencement Date APC Partners do not desire to enter into the LOC Sponsorship Agreement or the LOC does not conclude the LOC Sponsorship Agreement with the Commencement Date APC Partners by the deadline (a minimum of two (2) months from the day when the LOC contacts to such Commencement Date APC Partners) separately designated by the LOC, the LOC may discuss and enter into the LOC Sponsorship Agreement with any third parties other than such Commencement Date APC Partners.

10.2.3 If the APC newly enters into the agreements of sponsorship with the APC Partners after the Commencement Date, the APC shall promptly notify the LOC thereof. The LOC may discuss with such new APC Partners for entering into the LOC Sponsorship Agreement on a necessary basis. For the avoidance of doubt, such new APC Partners do not have the priority set forth in Clause10.2.2 in entering into the LOC Sponsorship Agreement.

10.3 Licensing Programme

- **10.3.1** The LOC is entitled to develop, work out and implement the Licensing Programme as one of the LOC Asian Para Games 2026 Marketing Programme upon the Licensing Rights granted in Clause 3.1.2 at LOC's sole discretion.
- **10.3.2** The LOC may enter into the license agreements (hereinafter referred to as, "**LOC License Agreement**") with respect to the rights of the Asian Para Games 2026 (including but not limited to the APC-IP and the LOC-IP) with the third parties (hereinafter referred to as, "**LOC Licensee**").
- 10.3.3 If the Licensed Products manufactured upon the LOC License Agreement between the LOC and the LOC Licensee inside the Region (but excluding the Host Country) are sold inside the Region (but excluding the Host Country), the amount of fifty (50) % (hereinafter referred to as, "NPC Distribution Money") of all net royalties (clear of all associated charges, VAT etc. from the royalties of the LOC License Agreement) received by the LOC will be distributed to the respective NPC which is located in the country that such Licensed Products are sold. For the avoidance of doubt,



if the Licensed Products upon the LOC License Agreement between the LOC and the LOC Licensee inside the Host Country are sold in the Region, the above provision of the NPC Distribution Money shall not be applicable.

10.4 Cooperation of JPSA

- **10.4.1** Upon the terms and conditions of this Agreement, the JPSA hereby approves that the APC and the LOC may commercially use and utilize (including the use and utilization by the LOC Partners, the suppliers, the broadcasting organisations and the LOC Licensee) the APC-IP and the LOC-IP in the Host Country, and the JPSA shall cooperate in the activities through the use and utilization by the LOC.
- **10.4.2** In addition to Clause 10.4.1, the JPSA hereby approves that the LOC may use (including the use by the LOC Partners, the suppliers, the broadcasting organisations and the LOC Licensee) the JPC Team Support Designation, free of charge, on a non-exclusive basis in the Host Country as part of the LOC Asian Para Games 2026 Marketing Programme upon the terms and conditions of the agreement separately concluded and executed between the JPSA and the LOC, and the JPSA shall cooperate in the LOC Asian Para Games 2026 Marketing Programme upon the discussion with the LOC.

10.5 Stadia Advertising

- 10.5.1 The LOC may set forth the provision of the Asian Para Games 2026 Venues which are free of publicity, advertising or sponsorship of products or companies that are not part of the LOC Partners (hereinafter referred to as, "Clean Stadia"), in the marketing plan (hereinafter referred to as, "Marketing Plan") set forth in Clause 10.6 on a necessary basis.
- **10.5.2** Commercial installations and advertising signs in support of the Asian Para Games 2026 are allowed at the Asian Para Games 2026 Venues in accordance with the provisions of the Marketing Plan.

10.6 Marketing Plan

No later than twelve (12) months after the Commencement Date, the LOC shall submit to the APC the first version of the Marketing Plan. The first version must contain all of the following elements for the Asian Para Games 2026. The APC shall make its advice on the first version and the subsequent versions to the LOC on a necessary basis.



- (a) Outline of the LOC Asian Para Games 2026 Marketing Programme (including outline of the Licensing Programme and the Ticket Programme)
- (b) Promotional plan outlining local and press promotional activities including volunteer promotions (if not included in the Media Plan as outlined in Clause11.6.1.1)
- (c) Outline of public relations and advertising of the Asian Para Games 2026

10.7 Ambush Marketing

- **10.7.1** The LOC agrees to take the necessary steps, at its cost, to prevent and/or terminate any Ambush Marketing or any unauthorised use of the Asian Para Games 2026 Commercial Rights in the Host Country. Specifically, if the LOC finds that any marketing, advertising or promotional plans in the Host Country organised by one (1) or more national federations, sports organisations or any other public or private entity (but excluding the existing and independent marketing activities made by the athletes and the Team, and the activities upon the JPC Marketing Programme) will have any impact on the LOC Asian Para Games 2026 Marketing Programme, and refer to or imply any connection with the Asian Para Games 2026 or any Team, the LOC shall approach such entity about ceasing such marketing, advertising or promotional plans.
- **10.7.2** In the event that the APC, the City and/or the JPSA become(s) aware of any Ambush Marketing or any unauthorised use of the Asian Para Games 2026 Commercial Rights by any third party (including through the report to the APC by the NPC), the APC, the City and/or the JPSA shall notify the LOC, without unreasonable delay, of such event, of the following details in particular:
 - (a) the identity of such the third party (if apparent);
 - (b) the nature of the activities;
 - (c) where the activities have taken place;
 - (d) the date(s) on which the activities were discovered; and
 - (e) any other information requested by the LOC (if apparent).
- **10.7.3** The APC agrees to take the necessary steps, at its cost, to prevent and/or terminate any unauthorised use of any rights of the APC other than the rights granted to the LOC upon Clause 3.1.2.



- **10.7.4** The JPSA agrees to take the necessary steps, at its cost, to prevent and/or terminate any unauthorised use of the JPC Team Support Designation.
- **10.7.5** Notwithstanding the provisions from Clause 10.7.1 to Clause 10.7.4, the APC and the LOC understand that the JPSA may grant the right of using the JPC Team Support Designation to the sponsors of the JPC Marketing Programme.

10.8 APC's Cooperation

The APC shall use its best efforts to undertake the following activities for performing the LOC Asian Para Games 2026 Marketing Programme outside the Host Country:

- (a) Obtaining each of the NPC's approval and assistance in all aspects with respect to exercising the Asian Para Games 2026 Commercial Rights in relation to the LOC Asian Para Games 2026 Marketing Programme in each jurisdiction area;
- (b) Obtaining each of the NPC's assistance for the countermeasures against the Ambush Marketing (including the protection of the APC-IP and the LOC-IP) in each jurisdiction area upon the LOC Asian Para Games 2026 Marketing Programme; and
- (c) Obtaining each of the NPC's approval for it that the particular LOC Partners specified in the LOC Asian Para Games 2026 Marketing Programme do the Sponsorship Activation through the use of the name of the Asian Para Games 2026, the Asian Para Games 2026 Emblem, the Asian Para Games 2026 Mascot and the Team Designation in accordance with the LOC Asian Para Games 2026 Marketing Programme.

11 Media Coverage

11.1 Games Broadcast Organisation

The LOC shall secure the establishment of a Games Broadcast Organisation (hereinafter referred to as, "**GBO**") determined by the LOC's sole discretion inside or outside the LOC. The GBO will be responsible for producing and distributing the broadcast footage of the Asian Para Games 2026 (hereinafter referred to as, "**Asian Para Games 2026 Broadcast Footage**"), and the provision of the services to the Rights Holding Broadcaster (hereinafter referred to as, "**RHB**") that have



obtained the broadcasting rights according to the Asian Para Games 2026 Media Services and Broadcast Requirements separately discussed and determined by the APC and the LOC. The method of the production and distribution of the Asian Para Games 2026 Broadcast Footage, and the sports and events for the Asian Para Games 2026 Broadcast Footage will be separately discussed and determined by the APC and the LOC.

11.2 Broadcasting Facilities and Services

The LOC shall make available to the RHB the technical facilities, support services and other requirements for the broadcasting of the Asian Para Games 2026, for a fee. The fee will be specified in the Rate Card after the consultation between the APC and the LOC.

11.3 GBO Broadcast Data

- **11.3.1** The copyrights (including the translation rights, adaptation rights and other rights stipulated in Copyright Act Article 27 of the Host Country and the rights of the original author in connection with the exploitation of a derivative work stipulated in Copyright Act Article 28 of the Host Country) of the Asian Para Games 2026 Broadcast Footage, broadcast signals or other material and data related to the Asian Para Games 2026 produced by, or on behalf of the GBO (hereinafter referred to as, collectively "**GBO Broadcast Data**") shall belong to the APC. The broadcaster will incorporate a suitable copyright notice at a minimum at the beginning and the end of each session of the daily coverage of each sport in a prominent position naming the APC as the copyright owner. Any rights other than the foregoing copyrights, related to the GBO Broadcast Data shall belong to the LOC.
- **11.3.2** The APC hereby agrees that the City will use the GBO Broadcast Data, free of charge, at the City's discretion, over the world for the purpose as defined in Clause 8.5.2 (a) and (b) after the dissolution of the LOC (for the avoidance of doubt, after the completion of the procedure of Chapter 4 Liquidation under Act on General Incorporated Associations and General Incorporated Foundations of the Host Country).
- **11.3.3** The Asian Para Games 2026 Broadcast Footage in possession of the LOC will be delivered by the LOC to the APC free of charge within ninety (90) days after the Closing Ceremony by the method set forth in this Clause. It shall be delivered as the digital copy in the hard disk drive of all Asian Para Games 2026 Broadcast Footage, including all venue footages, full



speed compilations of non-isolated slow-motion cameras and any other program or footage produced by the LOC or the LOC contractor. All the images should be accompanied by full and complete computerised time code logs. Sound must be synchronised with image. Furthermore, the LOC will deliver the Asian Para Games 2026 Broadcast Footage in possession of the LOC to the City free of charge at the dissolution procedure of the LOC (for the avoidance of doubt, the procedure of Section 6 Dissolution of Chapter 3 General Incorporated Foundations and Chapter 4 Liquidation under Act on General Incorporated Associations and General Incorporated Foundations of the Host Country) after the Asian Para Games 2026 in order the City to use it upon Clause 11.3.2.

11.3.4 Access to the Asian Para Games 2026 Broadcast Footage shall be given to the APC free of charge or license fees; however, the APC shall be prohibited from distributing the Asian Para Games 2026 Broadcast Footage to any third parties, but excluding the case set forth in Clause 11.4.

11.4 APC Web Broadcast Platforms

The LOC will ensure that the APC is given access to the Asian Para Games 2026 Broadcast Footage from the GBO and any other source under its control at any time free of charge or license fee. The GBO and/or the LOC will provide the necessary technical means, including a dedicated internet line of at least 50Mbit/s (upload) in total, for its exhibition on APC web broadcast platforms. However, the APC may make its exhibition of the Asian Para Games 2026 Broadcast Footage on the APC web broadcast platforms only upon the LOC's prior written approval to the extent that its exhibition will not hinder the exercise of the broadcasting rights of the LOC. The kind of sports and events exhibited on the APC web broadcast platforms will be discussed and determined by the APC and the LOC.

11.5 Media Facilities and Services

The LOC shall make available for the facilities, services and other requirements (hereinafter referred to as, "**Media Facilities and Services**") for the written and photographic press, the internet journalists and non-rights holding radio and television media in accordance with the Asian Para Games 2026 Media Services and Broadcast Requirements separately discussed and determined by the



APC and the LOC. The price (including the issue of a fee basis or free of charge basis) for the Media Facilities and Services will be separately discussed and determined by the APC and the LOC. The price of the Media Facilities and Services for a fee basis will be specified in the Rate Card. The LOC will submit the first version of a media operations plan to the APC no later than twelve (12) months prior to the Opening Ceremony.

11.6 Media Relations, Promotion and Public Awareness

11.6.1 Media Relations and Social Media

- **11.6.1.1** The LOC shall submit the first version of a detailed, timetabled plan to the APC for its written approval no later than twenty-four (24) months prior to the Opening Ceremony regarding media relations, digital and social media presence and public awareness (hereinafter referred to as, "Media Plan"). The goal of the Media Plan must be to raise the profile of the Asian Para Games 2026 and the athletes competing, locally and nationally. Such activity should encourage ticket sales, local and national media coverage and TV viewers.
- **11.6.1.2** The LOC shall include in the Media Plan how it anticipates it will use social media. In order to maximise impact, the LOC shall use the APC's social media channels, including, but not limited to: X (former name: Twitter), Facebook and Instagram: @asianparalympic, to promote the competition, in the build-up and for the Asian Para Games 2026 Period. If the LOC uses the APC's social media channels, the LOC shall contact the APC Media Manager who can approve access to the accounts and agree on the content and style of writing. The LOC shall abide by the Social Media Best Practice Guidelines separately discussed and determined by the APC and the LOC.
- **11.6.1.3** The official hash tag formats for the Asian Para Games 2026 shall be determined as follows:
 - (a) the official hash tag formats in English shall be determined through discussion between the APC and the LOC; and
 - (b) the official hash tag formats in any languages (including Japanese) other than English shall be determined by the LOC at its sole discretion.
- **11.6.1.4** The Media Plan shall also include details of how the LOC will engage with partner organisations to utilise their networks and maximize the reach of the Asian Para Games 2026, including, but not limited to, tourism



agencies, community groups, media, government agencies and the LOC Partners.

11.6.2 Asian Para Games 2026 Official Website

The LOC is responsible for delivering the Asian Para Games 2026 Official Website to be secured its accessibility at least twenty-four (24) months prior to the Opening Ceremony. The main content of the Asian Para Games 2026 Official Website will be discussed and determined by the APC and the LOC.

12 Obligations Subsequent to the Asian Para Games 2026

12.1 Report subsequent to the Asian Para Games 2026

- **12.1.1** Upon the conclusion of the Asian Para Games 2026, the LOC shall publish two (2) official reports (detailed and general) according to the detailed outlines provided by the APC, in English, no later than twelve (12) months after the Closing Ceremony. After their publication, three (3) electronic copies shall be delivered by the LOC to the APC free of charge in the format defined by the APC for their archives.
- 12.1.2 The LOC shall submit to the APC, for the APC's written opinion, a detailed outline of the contents of the official reports and shall submit the proofs of the contents of such official reports to the APC for the APC's prior written opinion before they are printed. The APC shall submit its written opinion for the detailed outline and the proofs to the LOC within thirty (30) days after the receipt thereof from the APC respectively.

12.2 Asian Para Games 2026 Information

The LOC shall record and maintain the following information relating to the Asian Para Games 2026 and provide the APC with it within twelve (12) months after the Closing Ceremony.

- (a) the number of athletes by nation and Sport Class
- (b) the number of medical-related persons by the Team
- (c) the number of sales of the Asian Para Games 2026 tickets
- (d) the number of sales for the Licensed Products and the number of persons registering with respect to the Asian Para Games 2026related services

12.3 Asian Para Games 2026 Designs and Promotional Materials



The LOC will provide the APC with the Asian Para Games 2026 Required Designs, the Asian Para Games 2026 Optional Designs, and the Promotional Materials, in the manner (e.g., paper copy, electronic copy) determined by the LOC within one (1) month after the Closing Ceremony.

12.4 Medals and Moulds

- 12.4.1 The LOC shall provide the APC with all the moulds for the Asian Para Games 2026 victory medals and all undistributed Asian Para Games 2026 victory medals, but excluding those provided to the City upon Clause 12.4.2 and the JPSA upon Clause 12.4.3 within one (1) month after the Closing Ceremony. In addition, within one (1) month after the Closing Ceremony, the LOC shall provide the APC with six (6) sets of the 2026 Asian Para Games victory medals (a total of eighteen (18) medals six (6) gold, six (6) silver and six (6) bronze). No reproductions or replicas of the medals or moulds of the Asian Para Games 2026 shall be made by the LOC without the prior written approval of the APC. The LOC shall cover the costs of transporting these moulds and medals to the APC.
- 12.4.2 Notwithstanding Clause 12.4.1, the LOC shall provide the City with two (2) sets of the moulds (a total of three (6) moulds two (2) gold, two (2) silver and two (2) bronze) and ten (10) sets of the Asian Para Games 2026 victory medals (a total of thirty (30) medals ten (10) gold, ten (10) silver and ten (10) bronze) within one (1) month after the Closing Ceremony. The City shall not use such moulds and medals for any purposes other than those set forth in Clause 8.5.2 (a) and (b).
- 12.4.3 Notwithstanding Clause 12.4.1, the LOC shall provide the JPSA with a set of the Asian Para Games 2026 victory medals (a total of three (3) medals one (1) gold, one (1) silver and one (1) bronze) within one (1) month after the Closing Ceremony. The JPSA shall not use such medals for any purposes other than the exhibitions in connection with the Asian Para Games 2026.

12.5 Knowledge Use

12.5.1 The JPSA and the LOC shall make available to the APC, free of charge, their knowledge and expertise in relation to planning, funding, organising and operating the Asian Para Games 2026, including availability of key documents, databases, statistical information, website content, as far as it is reasonable and does not violate the laws and regulations nor infringe upon any rights of any third parties.



12.5.2 The LOC also undertakes to contribute to the delivery of an APC debriefing of the Asian Para Games 2026 in the host city of the next edition of the Asian Para Games, by providing (and paying all expenses of) up to four (4) LOC staff one-time to attend the debriefing. The JPSA shall give its advice to the LOC's staff attending an APC debriefing on a necessary basis.

13 Representations and Warranties

13.1 Representations and Warranties of LOC

The LOC represents, warrants and undertakes that:

- (a) the LOC is free to enter into this Agreement, and its representatives are authorised to sign on it;
- (b) the LOC is duly organised and validly existing as the entity having the legal personality under the laws of the Host Country;
- (c) the LOC has the power and authority to perform the obligations set forth in this Agreement;
- (d) the conclusion and execution of this Agreement has been duly authorized by all necessary corporate actions of the LOC, and does not contravene the certificate of incorporation of the LOC; and
- (e) the LOC will at all times protect and maintain the value of the APC' rights granted by the APC in this Agreement and the APC-IP, and will conduct itself in a manner consistent with the high standard of care.

13.2 Representations and Warranties of APC

The APC represents, warrants and undertakes with that:

- (a) the APC is free to enter into this Agreement, and its representatives are authorised to sign on it;
- (b) the APC is duly organised and validly existing as the entity having the legal personality under the laws of United Arab Emirates;
- (c) the APC has the power and authority to perform the obligations set forth in this Agreement;
- (d) the conclusion and execution of this Agreement has been duly authorized by all necessary corporate actions of the APC, and does not contravene the Charter and the certificate of incorporation of the APC; and
- (e) the APC will respect the position of the City, the JPSA and the LOC to host the Asian Para Games 2026, and the APC will at all times



protect and maintain the value of the LOC-IP and will conduct itself in a manner consistent with the high standard of care.

13.3 Representations and Warranties of JPSA

The JPSA represents, warrants and undertakes that:

- (a) the JPSA is free to enter into this Agreement, and its representatives are authorised to sign on it;
- (b) the JPSA is duly organised and validly existing as the entity having the legal personality under the laws of the Host Country;
- (c) the JPSA has the power and authority to perform the obligations set forth in this Agreement;
- (d) the conclusion and execution of this Agreement has been duly authorized by all necessary corporate actions of the JPSA, and does not contravene the certificate of incorporation of the JPSA; and
- (e) the JPSA will at all times protect and maintain the value of the APC's rights granted by the APC in this Agreement, the APC-IP and the LOC-IP, and will conduct itself in a manner consistent with the high standard of care.

13.4 Representations and Warranties of the City

The City represents, warrants and undertakes that:

- (a) the City is free to enter into this Agreement, and its representatives are authorised to sign on it;
- (b) the City has the power and authority to perform the obligations set forth in this Agreement;
- (c) the conclusion and execution of this Agreement has been duly authorized by all necessary corporate actions of the City, and does not contravene any ordinances or rules of the City; and
- (d) the City will at all times protect and maintain the value of the APC's rights granted by the APC in this Agreement, the APC-IP and the LOC-IP, and will conduct itself in a manner consistent with the high standard of care.

14 Termination

14.1 Causes of Termination

14.1.1 Termination Right of APC



The APC shall be entitled to terminate this Agreement and withdraw the Asian Para Games 2026 at any time before the Asian Para Games 2026 in accordance with the procedure of Clause 14.2.1 if any of the following events has occurred or is occurring:

- (a) Breach: the LOC breaches any material obligation of this Agreement, any document referred to in this Agreement or any applicable law, and it is obvious that such breach would directly cause the objective impossibility of the host of the Asian Para Games 2026; or
- (b) Bankruptcy: the LOC becomes insolvent, or institutes (or there is instituted against it) proceedings in bankruptcy, dissolution, reorganisation or such similar proceedings, or makes an assignment for the benefit of creditors.

14.1.2 Termination Right of the City and LOC

The City and the LOC shall be entitled to terminate this Agreement and withdraw the Asian Para Games 2026 at any time before the Asian Para Games 2026 in accordance with the procedure of Clause 14.2.2 if any of the following events has occurred or is occurring:

- (a) Breach: the APC breaches any material obligation of this Agreement, any document referred to in this Agreement or any applicable law, and it is obvious that such breach would directly cause the objective impossibility of the host of the Asian Para Games 2026;
- (b) Bankruptcy: the APC becomes insolvent, or institutes (or there is instituted against it) proceedings in bankruptcy, dissolution, reorganisation or such similar proceedings, or makes an assignment for the benefit of creditors; or
- (c) Termination of the Host City Contract of the Asian Games 2026: the Host City Contract of the Asian Games 2026 is terminated and the Asian Games 2026 is withdrawn.

14.1.3 Termination by the Agreement of All Parties

All Parties may mutually agree on the termination of this Agreement in writing.

14.2 Procedure in Case of Termination

14.2.1 Termination by APC



The following procedure shall be used in case that the APC intends to terminate this Agreement and withdraw the Asian Para Games 2026 under Clause 14.1.1.

14.2.1.1 Occurrence of the Event of Clause 14.1.1(a):

- (a) if the APC determines that the event of 14.1.1 (a) has occurred or is occurring, the APC shall be entitled to put the LOC on notice in accordance with Clauses 16.6 and 16.7 (hereinafter referred to as, "Default Notice of 14.2.1.1") that the LOC is in default and to call upon the LOC to remedy the event of 14.1.1 (a) specified by the APC within sixty (60) days after the day of the LOC's receipt of the Default Notice of 14.2.1.1. Provided, however, if, on the day on which the Default Notice of 14.2.1.1 is sent by the APC to the LOC, the period remaining until the date of the Opening Ceremony is less than one hundred and twenty (120) days, the above-mentioned sixty (60) day period shall be reduced to one half of the number of days running from the date on which the Default Notice of 14.2.1.1 is sent until the date of the Opening Ceremony; and
- (b) if, following a Default Notice of 14.2.1.1 served in accordance with Clause 14.2.1.1 (a) above, the event of 14.1.1 (a) ascertained by the APC has not been remedied to the reasonable satisfaction of the APC within the period set forth in the said Clause 14.2.1.1 (a), the APC may terminate this Agreement and withdraw the Asian Para Games 2026 by giving the City, the JPSA and the LOC a termination notice in accordance with Clauses 16.6 and 16.7.

14.2.1.2 Occurrence of the Event of Clause 14.1.1(b):

if the APC determines that the event of 14.1.1 (b) has occurred or is occurring, the APC may immediately terminate this Agreement and withdraw the Asian Para Games 2026 by giving the City, the JPSA and the LOC a termination notice in accordance with Clauses 16.6 and 16.7.

14.2.2 Termination by the City and LOC

The following procedure shall be used in case that the City and the LOC intend to terminate this Agreement and withdraw the Asian Para Games 2026 under Clause 14.1.2.

14.2.2.1 Occurrence of the Event of Clause 14.1.2(a):

(a) if the City and the LOC determine that the event of 14.1.2 (a) has occurred or is occurring, the City and the LOC shall be entitled to put the APC on notice in accordance with Clauses 16.6 and 16.7 (hereinafter referred to as, "Default Notice of 14.2.2.1") that the



APC is in default and to call upon the APC to remedy the event of 14.1.2 (a) specified by the City and the LOC within sixty (60) days after the day of the APC's receipt of the Default Notice of 14.2.2.1. Provided, however, if, on the day on which the Default Notice of 14.2.2.1 is sent by the City and the LOC to the APC, the period remaining until the date of the Opening Ceremony is less than one hundred and twenty (120) days, the above-mentioned sixty (60) day period shall be reduced to one half of the number of days running from the date on which the Default Notice of 14.2.2.1 is sent until the date of the Opening Ceremony; and

(b) if, following a Default Notice of 14.2.2.1 served in accordance with Clause 14.2.2.1 (a) above, the event of 14.1.2 (a) ascertained by the City and the LOC has not been remedied to the reasonable satisfaction of the City and the LOC within the period set forth in the said Clause 14.2.2.1 (a), the City and the LOC may terminate this Agreement and withdraw the Asian Para Games 2026 by giving the APC and the JPSA a termination notice in accordance with Clauses 16.6 and 16.7.

14.2.2.2 Occurrence of the Event of Clause 14.1.2 (b) or (c):

if the City and the LOC determine that the event of 14.1.2 (b) or (c) has occurred or is occurring, the City and the LOC may immediately terminate this Agreement and withdraw the Asian Para Games 2026 by giving the APC and the JPSA a termination notice in accordance with Clauses 16.6 and 16.7.

14.3 Measures after Termination

- **14.3.1** In case that the APC has terminated this Agreement under Clause 14.1.1, the APC shall obtain the money which the LOC had paid to the APC by the time of the termination of this Agreement upon Clause 9.1 (Games Holding Fee). Furthermore, if the APC has suffered the damages exceeding the amount of this money, the APC may make its claim to compensate for such damages against the LOC.
- **14.3.2** In case that the City and the LOC have terminated this Agreement under Clause 14.1.2 (a) or (b), the APC shall immediately return to the LOC all the money which the LOC had paid to the APC by the time of the termination of this Agreement upon Clause 9.1 (Games Holding Fee). Each of the City and the LOC may make its claim to compensate for the damages suffered against the APC respectively.



- **14.3.3** In case that the City and the LOC have terminated this Agreement under Clause 14.1.2 (c), the measures after the termination of this Agreement shall be taken in accordance with Clause 15.4.
- **14.3.4** In case that All Parties have terminated this Agreement under Clause 14.1.3, the measures after the termination of this Agreement shall be determined through the discussion of All Parties.

15 Force Majeure

15.1 Non-Default by Force Majeure

If the performance of any obligation under this Agreement is prevented, restricted or interfered as a result of any event of Force Majeure as defined below, no Party shall be deemed to be in default under this Agreement by reason of such event (For the avoidance of doubt, the Party prevented, restricted or interfered as a result of such Force Majeure shall not be deemed as "Breaching Party" under Clause 16.5); provided, however, that the Party so affected shall give the other Party/Parties the prompt notice of such event of Force Majeure, use its best efforts to minimize, avoid or remove such event of non-performance, and continue performance hereunder without delay whenever such event is no longer applicable.

"Force Majeure" means:

- (a) the Host Country is at any time in a state of war or civil disorder, or is subject to an embargo decreed by the international community or in a situation officially recognized as one of belligerence;
- (b) earthquake, storm, tidal wave (tsunami), flood or any other natural disasters;
- (c) fire, terrorist act, disturbance or riot;
- (d) strike, lockout or any other labour trouble;
- (e) nuclear accident, breakdown of power or communication line, or computer malfunction;
- (f) occurrence or spread of contagious disease or infectious disease (e.g., influenza, SARS, COVID-19); and
- (g) any other events beyond the control of the Party.

15.2 Settlement of Force Majeure

If any Party under this Agreement acknowledges that any event of the



Force Majeure has occurred or is occurring and such event would cause any issues and problems of an adverse effect on the host, schedule, operation or financial plan of the Asian Para Games 2026, or the safety of the participants or the public in the Asian Para Games 2026, All Parties under this Agreement shall discuss such issues and problems, with a view to agreeing on resolution measures, including, but not limited to, the cancellation, relocation, advancement, postponement or interruption of the Asian Para Games 2026, an Asian Para Games 2026 Event or some Asian Para Games 2026 Events, and/or the negotiation in good faith in relation to additional costs and expenses. In the event of the occurrence of any disbursement of any additional costs and expenses, the APC, the City and the LOC shall bear such costs and expenses and the ratio of the charge for those shall be determined through the discussion among the APC, the City and the LOC.

15.3 Termination of this Agreement due to Force Majeure

The APC, the City or the LOC may terminate this Agreement and withdraw the Asian Para Games 2026 by giving the other Party/Parties a termination notice in accordance with Clause 16.6 and Clause 16.7, if All Parties under this Agreement have not reached the agreement to resolve the issues and problems arising out of the event of the Force Majeure through the discussion set forth in Clause 15.2 and it is obvious that:

- (a) such event would directly cause the objective impossibility of the host of the Asian Para Games 2026 in the City;
- (b) the safety of the participants or the public in the Asian Para Games 2026 would be seriously threatened or jeopardized directly due to such event;
- (c) such event would directly cause a material adverse effect on the operation of the Asian Para Games 2026 or the financial plan of the City and the LOC; or
- (d) such event has continued for the period of time longer than thirty (30) days.

15.4 Measures after Termination of this Agreement due to Force Majeure

In case of the termination of this Agreement by the APC, the City or the LOC under Clause 15.3, the measures after the termination of this Agreement shall be taken as follows:

15.4.1 Games Holding Fee of Clause 9.1



The LOC shall be exempt from the obligation of unpaid payments at the time of the termination of this Agreement upon Clause 15.3 among the payments set forth in Clause 9.1 (Games Holding Fee). If the LOC had paid any of the money set forth in Clause 9.1 (Games Holding Fee) to the APC by the time of the termination of this Agreement, the APC shall immediately return to the LOC the half of the amount of the money which the APC had actually received from the LOC upon Clause 9.1 (Games Holding Fee).

15.4.2 Costs and Expenses except for Games Holding Fee of Clause 9.1

15.4.2.1 Each of the APC, the City, the JPSA and the LOC shall bear any costs and expenses respectively in connection with the preparation and the host of the Asian Para Games 2026, which each of the APC, the City, the JPSA and the LOC has already disbursed by the time of such termination respectively.

15.4.2.2 The APC, the City and the LOC shall bear any costs and expenses arising out of or in connection with such termination other than the costs and expenses set forth in Clause 15.4.2.1, and the ratio of the charge for such costs and expenses to each of such Parties shall be determined through the discussion among such Parties.

15.4.3 Claim of Damages

No Party under this Agreement shall make any claims to compensate for the damages arising out of or in connection with such termination against the other Parties.

16 Miscellaneous

16.1 Language

All information and documentation submitted to the APC pursuant to this Agreement shall be conducted in English. The language of use for the Asian Para Games 2026 Period shall be English, and Japanese as the official language of the Host Country. In any case where Japanese and English versions of the same document exist, for all documents referred to in this Agreement, the English version shall prevail.

16.2 LOC's Responsibility

The LOC shall be responsible only for the obligations of the LOC stipulated under this Agreement and shall assume the legal and financial



responsibility within the scope of the budget of the LOC for the organisation and staging of the Asian Para Games 2026.

16.3 Commitment

The City and the LOC make the commitments that no agreements inconsistent with the terms and conditions under this Agreement will be entered into between, on the one hand, the City and/or the LOC and, on the other hand, any third parties without the approval of the APC.

16.4 Confidentiality

16.4.1 "Confidential Information" means:

- (a) the negotiation process (including the contents of any drafts of this Agreement, and any comments and discussion on such drafts), the conclusion and the contents of this Agreement;
- (b) any consultation, discussion, negotiation, commitments, agreements and contracts related to this Agreement;
- (c) any contractual documentation related to this Agreement (including any associated appendices to this Agreement) entered into between the Parties or All Parties; and
- (d) any and all information, including, but not limited to financial information, technical information, data, documents and know-how disclosed by one Party (hereinafter referred to as, "Disclosing **Party**") to any other Party/Parties (hereinafter referred to as, "Receiving Party") in connection with the negotiation, conclusion and performance of this Agreement, whether by tangible or intangible media, and (i) in the case of a written disclosure or on other tangible media, which shall be marked "Confidential" or the like, and (ii) in the case of an oral or visual disclosure, which shall be identified as confidential at the time of the disclosure and given to the Receiving Party by the Disclosing Party a written notice (including a notice by email or facsimile) to the effect that the information is the Confidential Information within thirty (30) days after such disclosure (in the context of this Paragraph (d), the Confidential Information disclosed electronically (including by email) by the Disclosing Party to the Receiving Party shall be deemed to be disclosed in tangible media).
- **16.4.2** Each Party under this Agreement undertakes to keep confidential the Confidential Information and shall not disclose the Confidential



Information to any third parties (other than All Parties under this Agreement), except to the extent that:

- (a) the disclosure of the information is required by relevant laws, ordinances or court orders;
- (b) the disclosure of the information is necessary in the judicial proceedings and the contents and scope of such disclosed information, and the details of such proceedings shall be, in advance, notified to the other Parties (in case of the Confidential Information (a)(b)(c)) or the Disclosing Party (in case of the Confidential Information (d));
- (c) the information is in the public domain (but excluding the case that it has entered the public domain by a breach of this Clause 16.4);
- (d) the disclosure of the information is made to the APC's, the City's, the JPSA's or the LOC's professional advisers (e.g., legal counsels and accountants), auditors, licensees, suppliers, contractors, other persons or other entities who have a legitimate need to know such information in connection with the preparation and host of the Asian Para Games 2026 and who have a professional obligation of confidentiality or agree in advance to be bound by the provisions of this Clause 16.4.2 in writing, to the extent that such disclosure shall be necessary ; and
- (e) the information is disclosed upon the consent of the other Parties (in case of the Confidential Information (a)(b)(c)) or the Disclosing Party (in case of the Confidential Information (d)).
- **16.4.3** Save where otherwise provided for in this Agreement or required by law, this Agreement and/or any part or parts therein (including any appendices to this Agreement) shall not be disclosed to the public by the Parties herein, provided always that this Agreement translated in the Japanese language only be and is allowed to be disclosed to the public through its official website etc.

16.5 Indemnification

If any Party of the APC, the City or the LOC (hereinafter referred to as, "**Breaching Party**") makes any breach, non-performance or violation (hereinafter collectively referred to as, "**Breach**") of this Agreement due to the reason attributable to the Breaching Party and such Breach causes any damage to the non-breaching Party/Parties (hereinafter referred to as, "**Non-Breaching Party**"), the Breaching Party shall compensate the Non-Breaching Party for the damage incurred by the Breach. The damage



to be indemnified shall not include any indirect, special, incidental, punitive or consequential damage.

16.6 Notices

- **16.6.1** Unless otherwise specified in this Agreement, all notices, approvals, consents or other communications (hereinafter collectively referred to as, "**Notices**") in connection with this Agreement shall be in writing and shall be given to the Party if delivered personally or sent by registered airmail, international courier service or facsimile at the addresses specified in Clause 16.7.1. In the case of facsimile, the notifying Party shall make the prompt confirmation by registered airmail or international courier service.
- **16.6.2** Notwithstanding Clause 16.6.1, in the event that the notifying Party and the notified Party separately agree to give some of the Notices by e-mail through discussion, the notifying Party may give such Notices by transmitting to the e-mail address of the notified Party set forth in Clause 16.7.1 the writing of such Notices as attached file.

16.7 Addresses

16.7.1 The nominated recipient, address, facsimile number and e-mail address of each Party is:

Asian Paralympic Committee Nominated recipient:

Tarek Souei, Chief Executive Officer **Address:** P.O. Box 232111, 3/F Dubai Club for People of Determination, Al Nahda St., Al Qusais area, Dubai, United Arab Emirates **Facsimile:** 971-4-298-8459 **E-mail:** tarek.souei@asianparalympic.org

City (Aichi Prefecture)

Nominated recipient: Hideaki OHMURA, Governor Copy: Kenji TAGAMI, Director



Asian Games and Asian Para Games Division, Bureau of Sports Address: 3-1-2 Sannomaru, Naka-Ku, Nagoya, Aichi 460-8501 Japan Facsimile: 81-52-951-1006 E-mail: ag@pref.aichi.lg.jp

City (City of Nagoya)

Nominated recipient: Takashi KAWAMURA, Mayor Copy: Koichi NAGAMACHI, Director Asian Games & Asian Para Games Promotion Office, General Coordination Department, General Affairs Bureau Address: 1-1 Sannomaru, 3-Chome, Naka-Ku, Nagoya, Aichi 460-8508 Japan Facsimile: 81-52-972-4205 E-mail: a2231@somu.city.nagoya.lg.jp

Japanese Para Sports Association

Nominated recipient: Masaki FUJIWARA, Senior Executive Director Copy: Shinji NAKAMAE, Senior Manager Performance Development Section, Performance Development Department Address: 2-13-6 Nihonbashi-Kakigara-Cho, Chuou-Ku, Tokyo 103-0014 Japan Facsimile: 81-3-5641-1213 E-mail: m-fujiwara@parasports.or.jp

<u>Aichi-Nagoya Asian Games and Asian Para Games Organizing</u> <u>Committee</u> Nominated recipient:



Satoshi MURATE, Secretary General Copy: Kei KOJIMA, Director General Address: 3-2-1 Sannomaru, Naka-Ku, Nagoya, Aichi 460-0001 Japan Facsimile: 81-52-746-9150 E-mail: ainagoc@aichi-nagoya.2026.org

16.7.2 If each Party desires to change its addresses specified in Clause 16.7.1, the Party shall notify the other Parties of such change in writing.

16.8 Assignment

Each Party shall not assign any rights or obligations under this Agreement to any third parties without the prior written approval of the other Parties.

16.9 Entire Agreement and Amendment

- **16.9.1** From the Commencement Date, this Agreement constitutes the entire understanding between All Parties and supersedes all prior negotiations, understandings and agreements between the Parties or All Parties related to its subject matter.
- **16.9.2** Any amendments to this Agreement shall be valid only if made in writing and signed by the respective authorized representative of the APC, the City, the JPSA and the LOC.

16.10 No relationship

All Parties have no authority to bind each other to any contract and no Party shall give the impression that it may act as a representative, agent or attorney of another Party. Nothing contained in this Agreement shall be construed or deemed to make the Parties or All Parties a joint venture or partners.

16.11 Undue Hardship



Should any provision of this Agreement give rise to undue hardship affecting the City and/or the LOC because of the substantial change of economic or political circumstances after the Commencement Date, the City and/or the LOC may request the APC to consider such changes as may be reasonable in the circumstances. The APC, and the City and/or the LOC shall determine the change of such provision of this Agreement through the discussion, considering the budget in the financial plan and any other factors of the Asian Para Games 2026.

16.12 Severability

If a provision of these terms and conditions is determined unlawful and/or unenforceable, the other provisions shall continue with full effect. If the unlawful or unenforceable provision would be lawful or enforceable if part of it were deleted, that part shall be deemed deleted and the remainder of the provision shall remain in effect.

16.13 Anti-Corruption

All Parties agree to desist from all practices which may lead to penal liability due to fraud, embezzlement, insolvency crimes, crimes in violation of competition, guaranteeing advantages, bribery, acceptance of bribes or other corruption crimes on the part of persons employed by the Party, or other third parties.

16.14 Scope of Performance of Obligations

Each Party shall perform its obligations in accordance with the provisions under this Agreement within the scope specified in the provisions unless otherwise separately agreed by the Party.

16.15 Responsibility for JPC's Obligations

The JPSA undertakes to, by its responsibilities, cause the JPC to perform the JPC's obligations set forth in this Agreement and abide by the JPC's commitments herein, and cause the JPC not to act against any provisions in this Agreement. If the JPC fails to perform the obligations or abide by the commitments, or acts against any provisions in this Agreement, it shall be deemed as the breach by the JPSA in this Agreement and the



JPSA shall be responsible for it in accordance with the provisions of this Agreement.

16.16 Governing Law and Dispute Resolution

- **16.16.1** This Agreement shall be governed by and construed in accordance with the laws of Japan.
- **16.16.2** All disputes arising out of or in connection with this Agreement shall be finally settled by the arbitration under the Rules of Arbitration of the International Chamber of Commerce. The number of arbitrators shall be three (3). In the event of two (2) Parties to the arbitration, each Party shall nominate one (1) arbitrator. In the event that the Claimant and/or the Respondent are/is multiple to the arbitration, the Claimant (s) shall jointly nominate one (1) arbitrator. The third arbitrator shall be jointly nominate one (1) arbitrator. The third arbitrator shall be jointly nominate one (1) arbitrator. The third arbitrator shall be jointly nominate one (1) arbitrator. The third arbitrator shall be jointly nominated by these two (2) arbitrators. The arbitration shall take place in Nagoya-City, Japan. The language of the arbitration shall be English. The award thereof shall be final and legally binding upon All Parties/the Parties with respect thereto.

16.17 Appendices

The following Appendices are attached hereto and form an integral part of this Agreement. All Parties agree to be bound by the contents of all the following Appendices:

Appendix A: Definitions and Interpretation Appendix B: Accessibility Standard Appendix C: Sports Programme of Asian Para Games 2026 Appendix D: APC Family Appendix E: Details of Games Holding Fee Appendix F: Commencement Date APC Partners

End of the page



IN WITNESS WHEREOF, All Parties have caused this Agreement to be executed in quintuplicate, each of which shall be considered an original, by their respective representatives thereunto duly authorized.

Asian Paralympic Committee (APC)

Signature: Printed Name: Majid Rashed Position: President

Signature: Printed Name: Tarek Souei Position: Chief Executive Officer

Aichi Prefecture

Signature: Printed Name: Hideaki OHMURA Position: Governor



City of Nagoya

Signature: Printed Name: Takashi KAWAMURA Position: Mayor

Japanese Para Sports Association (JPSA)

Signature: Printed Name: Kazuyuki MORI Position: President

Aichi-Nagoya Asian Games and Asian Para Games Organizing Committee (LOC)

Signature: Printed Name: Satoshi MURATE Position: Secretary General



Appendix A

Definitions and Interpretation

1. Definition

In this Agreement, the capitalized terms shall have the meanings set forth in this Appendix A or as otherwise set forth in this Agreement.

"Accessibility Consultant": as defined in Clause 6.1.5

"Accessibility Standard": as defined in Clause 2.4.3

"AF": the international federations based in the Region responsible for administrating the Scheduled Sports on national level and recognized by the APC (For the avoidance of doubt, the international federations administrating any sports other than the Scheduled Sports are not included in "AF")

"Agreement": Aichi-Nagoya 2026 Asian Para Games Host City Agreement

"All Parties": all parties among the APC, the City, the JPSA and the LOC

- "Ambush Marketing": any activity, commercial or non-commercial, undertaken by any person or entity, whether public or private, that creates, implies or refers to a direct or indirect association of any kind (an association easily in the minds of members of the public) with the APC, the LOC or the Asian Para Games 2026, which has not been properly authorized by the APC, the LOC, the City or the JPSA
- "APC": Asian Paralympic Committee

"APC Family": as defined in Appendix D

"APC Partners": the third parties with whom the APC has entered into the agreements of sponsorship upon the APC International Marketing Programme (the APC's own International Marketing Programme to generate financial, technical and promotional support from international corporations etc. for the benefit of the entire Asian and global Paralympic Movement)

"APC/IPC Regulations and Requirements": as defined in Clause 2.3

"APC-IP": as defined in Clause 8.1.1

"Asian Games 2026": Asian Games in and around the City governed by the Olympic Council of Asia (OCA) in 2026



- **"Asian Para Games"**: multi-competition of Para sports in the Region governed by the APC for the purpose of the promotion of developing the Paralympic Movement and Para sport events in the Region
- "Asian Para Games 2026": the Asian Para Games (Official Name: Aichi-Nagoya 2026 Asian Para Games) to be held principally in and around the City for the Asian Para Games 2026 Period (including the Asian Para Games 2026 Event and the Asian Para Games 2026-Related Events for the Asian Para Games 2026 Period, but excluding the Host City Events, the Test Event, the Education Programme, the Athlete Education Programme and the Para Sports Development Programme), under the auspices of the APC
- **"Asian Para Games 2026 Athletes' Village"**: any facilities (including hotels) designated by the LOC as the appropriate accommodation for the athletes and NPC Team Officials
- "Asian Para Games 2026 Broadcast Footage": as defined in Clause 11.1
- "Asian Para Games 2026 Budget": plan related to income and expenditure of the Asian Para Games 2026
- "Asian Para Game 2026 Commercial Revenues": as defined in Clause 3.1.3
- "Asian Para Game 2026 Commercial Rights": as defined in Clause 3.1.1
- "Asian Para Games 2026 Events": the Scheduled Sports and the sport events related to the Scheduled Sports (including, but not limited to, the Scheduled Sports and the Victory Ceremonies performed before the Opening Ceremony) to be held as part of the Asian Para Games 2026
- "Asian Para Games 2026 Marks": as defined in Clause 8.3.1
- "Asian Para Games 2026 Optional Designs": as defined in Clause 8.3.4
- "Asian Para Games 2026 Partner Categories": the categories on the basis of business type etc. set up by the LOC for avoiding the conflict of the interests of the LOC Partners when the LOC enters into the LOC Sponsorship Agreement with the LOC Partners
- "Asian Para Games 2026 Performance Record": as defined in Clause 5.4
- **"Asian Para Games 2026 Period"**: the period from seven (7) days before the Opening Ceremony to two (2) days after the Closing Ceremony
- "Asian Para Games 2026 Photograph": as defined in Clause 5.5.1
- "Asian Para Games 2026-Related Events": the events or activities officially organised, supported, sanctioned by, or staged under the auspices of the APC or the LOC, and which are designed to promote, celebrate, enhance or facilitate the hosting and staging of the Asian Para Games 2026, including the Opening



Ceremony, the Closing Ceremony, and other events related to the Asian Para Games 2026 under the auspice of the LOC, but excluding the Asian Para Games 2026 Events

"Asian Para Games 2026 Required Designs": as defined in Clause 8.3.2

- "Asian Para Games 2026 Site": (i) any Asian Para Games 2026 Venues, (ii) any training site, (iii) any official parking area, accreditation centre, ticketing centre (if installed only for the Asian Para Games 2026), volunteer centre, the international broadcast centre (if located within the City), and/or any hospitality area (including hospitality boxes and hospitality villages), used in relation to the Asian Para Games 2026 and (iv) any other locations, areas and/or facilities designated by the APC as an official site located within the City in relation to the Asian Para Games 2026, and to which access is permitted only to the GIAC holders or ticket holders or any other officially-sanctioned ticket or other access device
- "Asian Para Games 2026 Venue": any venue, facility, site and stadium at, in or through which the Asian Para Games 2026 Events are held, including any permanent or temporary facilities or areas, to which a ticket or a GIAC card is required in order to gain access
- **"Athlete Education Programme"**: programme implemented by the LOC upon Clause 5.20.3
- "Breach": as defined in Clause 16.5
- "Breaching Party": as defined in Clause 16.5
- "Capitation Fee": as defined in Clause 9.3.1
- "City": Aichi Prefecture and City of Nagoya
- "Clean Stadia": as defined in Clause 10.5.1
- **"Closing Ceremony"**: the closing ceremony held to mark the conclusion of the Asian Para Games 2026
- "CoComs": as defined in Clause 7.1.1
- "Competition Sessions": the time period for which the competition is held
- "Commencement Date": as defined in Clause 1
- "Commencement Date APC Partners": as defined in Clause 10.2.2
- **"Community Programme"**: programme implemented by the City upon Clause 5.21.1
- "Default Notice of 14.2.1.1": as defined in Clause 14.2.1.1



- "Default Notice of 14.2.2.1": as defined in Clause 14.2.2.1
- "Disclosing Party": as defined in Clause 16.4.1(d)
- **"Education Programme"**: programme implemented by the LOC upon Clause 5.20.2
- "Fee for Administrating, Planning and Advising": the fee for which the APC advises the LOC in relation to organising, planning and staging the Asian Para Games 2026 upon Clause 2.1.6, the fee for which the APC holds the CoComs, supervises and supports the LOC upon Clause 7.1, and the fee for which the APC supports (including advice) the LOC upon Clauses 5.11.3, 5.11.4, 5.13.2, 5.14, 6.1.5, 10.6 and other Clauses
- **"Fee for Granting the Asian Para Games 2026 Commercial Rights"**: the fee for the APC's granting the Asian Para Games 2026 Commercial Rights to the LOC upon Clause 3.1
- "Financial Plan": as defined in Clause 9.5.1

"FOP": field of play

- "Force Majeure": as defined in Clause 15.1
- "Games Holding Fee": the fee which the LOC shall pay to the APC in consideration of the LOC's organising and hosting the Asian Para Games, an exclusive property of the APC, by use of certain commercial rights and services provided by the APC, in and around the City of the Host Country, upon Whereas Clause 1. The details of the Games Holding Fee are defined in Clause 9.1.1 and Appendix E
- "Games Officials Hotel": as defined in Clause 6.5.1
- "Games Sanctioning Fee": the fee which the LOC shall pay to the APC in consideration of the APC's approval for the LOC's organising and hosting the Asian Para Games 2026, in and around the City of the Host Country, upon Whereas Clause 3
- "GBO": as defined in Clause 11.1
- "GBO Broadcast Data": as defined in Clause 11.3.1
- "GIAC": as defined in Clause 5.1.1
- "Host City Events": any events or activities, under the auspice of the City or any third party acting with the authority of the City, which are designed to promote, celebrate, enhance or facilitate the hosting and staging of the Asian Para Games 2026 (including Community Programme)

"Host Country": Japan



- **"IF"**: the international federations responsible for administrating the Scheduled Sports on world or national level and recognized by the APC (For the avoidance of doubt, the international federations administrating any sports other than the Scheduled Sports are not included in "IF")
- **"IF Rules"**: collective terms of "IF rules and regulations" and "the IF classification rules and regulations" as defined by each IF managing the Scheduled Sports
- "IPC": International Paralympic Committee
- "IPC Information Systems": as defined in Clause 5.1.2
- **"IPC Worldwide Paralympic Partners"**: the sponsors of the highest rank in the sponsorship programme of the IPC
- "ISTI": as defined in Clause 5.18.1
- "JPC": Japanese Paralympic Committee
- "JPC Marketing Programme": the contents of the plan with respect to any and all marketing activities performed by the JPC (including the JPC Official Sponsorship Programme)
- **"JPC Team Support Designation"**: a support designation connoting a relationship with the JPC's Team for Aichi-Nagoya 2026 Asian Para Games (Asian Para Games 2026)
- "JPSA": Japanese Para Sports Association
- **"Licensed Products"**: the products commercialized in relation to the Asian Para Games 2026 upon the Licensing Rights
- "Licensing Programme": the contents of the plan to manufacture and sell the products related to the Asian Para Games 2026 by entering into the license agreements with the third parties
- "Licensing Revenues": any and all revenues (including VIK) generated from the Licensing Rights in any and all manners
- "Licensing Rights": the rights (including the rights to manufacture and sell the Licensed Products upon the license agreements) to develop, manufacture and sell the products related to the Asian Para Games 2026 upon the rights (including the APC-IP, the LOC-IP) of the Asian Para Games 2026
- "LOC": Aichi-Nagoya Asian Games and Asian Para Games Organizing Committee
- "LOC Asian Para Games 2026 Marketing Programme": as defined in Clause 10.1.1
- **"LOC License Agreement"**: the license agreement which the LOC has entered into with the LOC Licensee upon the Licensing Rights



- **"LOC Licensee"**: the licensee with whom the LOC has entered into the LOC License Agreement upon the Licensing Programme
- "LOC Marketing Agent": as defined in Clause 10.1.2
- **"LOC Partners"**: the third parties with whom the LOC has entered into the LOC Sponsorship Agreement
- **"LOC Sponsorship Agreement"**: the agreement with certain specified third party to authorize such third party to use certain value of the Asian Para Games 2026 as the sponsors by the LOC, including providing the advertising opportunities at the Asian Para Games 2026 Venues and providing the hospitality upon the Sponsorship Rights etc.
- "LOC-IP": as defined in Clause 8.1.2
- **"Look"**: design decoration made on the advertisement, board, sign, media back board etc.
- **"Look Material"**: material for decorating the City and incorporating and reflecting the Asian Para Games 2026 Marks throughout the City
- "Marketing Plan": as defined in Clause 10.5.1
- "Marketing Revenues": any and all revenues (including VIK) generated from the Marketing Rights in any and all manners
- "Marketing Rights": in any and all media, throughout the universe, and in all languages, any and all advertising rights, promotional rights, rights of endorsement, official sponsor designation rights, joint marketing rights, premium and giveaway rights, Licensing Rights, creation and use right of intellectual property rights (including creation and use of the LOC-IP, use of the APC-IP), catering and concession rights, Sponsorship Rights, hospitality rights, travel and tourism rights, accommodation rights, transportation rights, publishing rights, betting/gaming rights, retailing rights, music rights, philatelic rights, numismatic rights, lottery rights, auction rights and any other rights and/or associated commercial opportunities (whether now known or hereafter invented) relating to the Asian Para Games 2026, to the extent that such rights are not the Media Rights and the Ticket Rights
- "Master Plan": as defined in Clause 2.8
- "Media Facilities and Services": as defined in Clause 11.5
- "Media Plan": as defined in Clause 11.6.1.1
- "Media Revenues": any and all revenues (including VIK) generated from the Media Rights in any and all manners



- "Media Rights": the rights (including broadcasting rights), throughout the universe, and in all languages, to report upon, record, transmit or otherwise exploit any still or moving visual-only images, any audio-only material, any audio-visual material, any text and any data with respect to the Asian Para Games 2026, by any means whatsoever (whether now known or hereafter invented), any aspect or element of the Asian Para Games 2026 Events on a live, as-live and/or delayed basis, in full length or in part, in any media and by any means of transmission or delivery whether now known (including successor technologies) or hereafter invented. For the avoidance of doubt, the right to broadcast, stream and/or transmit the basic audiovisual feed (or any supplemental feed) and the right to transmit radio commentary of any Asian Para Games 2026 Events constitute the Media Rights. The Media Rights include the right to record, create and exploit any official films or any official concerts of the Asian Para Games 2026 and/or similar audiovisual products and programming and include the fixed media rights, public exhibition rights and the in-flight rights. Furthermore, the Media Rights contain the rights of the Asian Para Games 2026 Performance Record, the rights of the Asian Para Games 2026 Photograph and the rights of the GBO Broadcast Data
- "Non-Breaching Party": as defined in Clause 16.5
- "Notices": as defined in Clause 16.6
- "NPC": National Paralympic Committees
- "NPC Distribution Money": as defined in Clause 10.3.3
- "NPC Team Officials": the officials (including managers and support staff) accompanying the Team of each NPC in the Asian Para Games 2026
- **"Opening Ceremony"**: the opening ceremony held to mark the opening of the Asian Para Games 2026
- "Operational Requirements of Clause 2.4.1": as defined in Clause 2.4.1
- "Operational Requirements of Clause 2.4.2": as defined in Clause 2.4.2
- "Other Marketing Revenues": any and all revenues (including VIK) generated from Other Marketing Rights in any and all manners
- "Other Marketing Rights": the rights, excluding the Sponsorship Rights and the Licensing Rights among the Marketing Rights
- **"Para Sport Development Programme"**: programme implemented by the APC, the JPSA, the City and /or the LOC upon Clause 5.21.2
- "Paralympic Sports": Para sports held in Paralympic games
- "Parties": two or more parties among the APC, the City, the JPSA and the LOC



- "Party": one party among the APC, the City, the JPSA and the LOC
- "Pictograms": a picture symbolizing the sports or events of the Scheduled Sports
- **"Promotion Materials"**: advertisements used to promote the Asian Para Games 2026
- **"Rate Card"**: price list of the technical equipment, support service and other necessities which the LOC provides to the media for a fee
- "Receiving Party": as defined in Clause 16.4.1 (d)
- "Region": the region of the Asia
- "RHB": as defined Clause 11.1
- "Scheduled Sports": as defined Clause 4.2.1
- "SDMS": as defined in Clause 5.1.2
- **"Sponsorship Activation"**: sponsor's activities (e.g., advertisement, promotion) by using the rights granted upon the LOC Sponsorship Agreement related to the Asian Para Games 2026
- **"Sponsorship Revenues"**: any and all revenues (including VIK) generated from the Sponsorship Rights in any and all manners
- "Sponsorship Rights": the right to authorize certain specified third party to use certain value of the Asian Para Games 2026 as the sponsors, including using the Official Sponsor Designation of the Asian Para Games 2026, providing the advertising opportunities at the Asian Para Games 2026 Venues, providing the hospitality and providing the social contribution through the promotion of the development of the Paralympic Movement and Para sports
- "STMOU": as defined in Clause 2.7
- "**Team**": the teams of athletes, NPC Team Officials, representing each of the NPC in the Asian Para Games 2026
- **"Team Designation"**: a designation connoting a relationship with the NPC (excluding the JPC) and/or its Team (e.g., (1) Official Partner, Sponsor and Supplier of the Team of [country], (2) Support of the Team of the NPC (excluding the JPC)
- **"Team Welcome Ceremony"**: team welcome ceremony of the Asian Para Games 2026
- "Technological Services": as defined in Clause 5.11.1



- **"Test Events"**: any events performed by the LOC or the third parties upon approval of the LOC before the Asian Para Games 2026 Period for the purpose of enhancing the ability of management of the Asian Para Games 2026
- **"Ticket Programme"**: the contents of the plan to sell or to have any third parties sell the admission tickets of the Asian Para Games 2026
- **"Ticket Revenues"**: any and all revenues (including VIK) generated from the Ticket Rights in any and all manners
- "Ticket Rights": the rights to sell or to have any third parties sell the admission tickets of the Asian Para Games 2026
- "Torch": torch used in torch relays of the Asian Para Games 2026
- "TUE": therapeutic use exemptions
- "Victory Ceremonies": ceremonies to award victory medals and diplomas etc. of the Asian Para Games 2026

"VIK": as defined in Clause 3.1.3

"WADA": as defined in Clause 5.18.1

2. References and Other Aids to Construction

Unless the context otherwise requires, this Agreement is to be interpreted according to the following:

- (a) words importing the singular include the plural and vice versa, words importing the masculine gender include the feminine and neuter and vice versa;
- (b) references to persons include incorporated and unincorporated bodies or partnerships and those persons' legal representatives, successors and permitted assigns;
- (c) references to "day" shall mean a calendar day;
- (d) reference to any provision of law includes any amendment, repromulgation or supplement to the provision;
- (e) references to "include", "including" and similar are to be construed without limitation;
- (f) headings of this Agreement are for convenience only and shall not affect the interpretation of this Agreement;
- (g) "writing", and any cognate expression, include paper media and electronic media (e.g., PDF data);



- (h) a statute or a provision of a statute is a reference to that statute or provision as amended or re-enacted at the relevant time;
- (i) "this Agreement" is a reference to this Host City Agreement of Asian Para Games 2026 and the Appendices, as amended or supplemented (where and as permitted hereunder) at the relevant time;
- (j) an Appendix is a reference to any appendix to this Agreement; and
- (k) a paragraph, a section or a Clause is a reference to a provision of this Agreement (other than an Appendix).



Appendix B

Accessibility Standard

- 1. The LOC will aim for it that the accessibility of the Asian Para Games 2026 Venues will be in conformity with the standard set forth in "the Aichi Prefecture Ordinance on the Promotion of Urban Development with Accessibility of Facilities for All People (*HITO NI YASASHII MACHI ZUKURI NO SUISHIN NI KANSURU JOREI*)".
- 2. The LOC will not take large-scale measures requiring the renovation of the building frame in the Asian Para Games 2026 Venues and the Asian Para Games 2026 Athletes' Village utilizing the existing facilities.
- 3. The LOC will make its efforts to ensure the appropriate accessibility of the areas of the Asian Para Games 2026 Venues used by the athletes and the Asian Para Games 2026 Athletes' Village with consideration of the characteristics of each sport and the disability situation of the athletes and NPC Team Officials, referring to the record of National Sports Festival for People with an Impairment (*"ZENKOKU SYOGAISYA SPORT TAIKAI"*) and International Sports Competition for People with Impairment of single sport previously held in the Host Country.
- 4. If additional wheelchair seating is required in the Asian Para Games 2026 Venues, the LOC will consider take the measures with less burden, such as providing temporary seating in the side of Field of Play ("FOP").



Appendix C

| No. | Code | Sport | |
|-----|------|-------------------|--|
| 1 | AR | Para Archery | |
| 2 | AT | Para Athletics | |
| 3 | BA | Para Badminton | |
| 4 | BO | Boccia | |
| 5 | CY | Para Cycling | |
| 6 | FB | Blind Football | |
| 7 | GB | Goalball | |
| 8 | JU | Judo | |
| 9 | PO | Para Powerlifting | |

Sports Programme of Asian Para Games 2026

| No. | Code | Sport | |
|-----|------|-----------------------|--|
| 10 | SH | Shooting Para Sport | |
| 11 | SV | Sitting Volleyball | |
| 12 | SW | Para Swimming | |
| 13 | TT | Table Tennis | |
| 14 | ТК | Para Taekwondo | |
| 15 | WB | Wheelchair Basketball | |
| 16 | WF | Wheelchair Fencing | |
| 17 | WR | Wheelchair Rugby | |
| 18 | WT | Wheelchair Tennis | |



Appendix D

APC Family

APC Family consists of the following members.

| Members | | | |
|---|--|--|--|
| APC Executive Board Members | | | |
| APC Standing Committee Chairpersons | | | |
| Guides/Assistants/Translators for APC Executive Board and Standing Committee Members | | | |
| APC Executive Board Accompanying Guests | | | |
| APC Standing Committee Members | | | |
| APC Staff | | | |
| NPC Presidents | | | |
| NPC Secretary-Generals | | | |
| Guests from IPC | | | |
| Guests from Regional Organisation | | | |
| Guests from IF | | | |
| Guests from International Organisations of Sports for the Disabled (IOSD) | | | |
| APC Guests | | | |
| Guests from Future Organising Committee | | | |



<u>Appendix E</u>

Details of Games Holding Fee

The details of the Games Holding Fee are as follows.

| Games Holding Fee | | | | | |
|---|---------------------------|---|--|--|--|
| | Details | Amount | | | |
| Games Sanc | tioning Fee | Two Million Three Hundred Thousand US Dollars (USD 2,300,000) | | | |
| Fee for Administrating, Planning and Advising | | Three Hundred Thousand US Dollars (USD 300,000) | | | |
| Fee for | Sponsorship Rights | Two Hundred Eighty Thousand US Dollars (USD 280,000) | | | |
| Granting the Asian | Licensing Rights | Free of Charge | | | |
| Para Games | Other Marketing Rights | Forty Thousand US Dollars (USD 40,000) | | | |
| 2026 Commercial | Media Rights | Forty Thousand US Dollars (USD 40,000) | | | |
| Rights | Ticket Rights | Forty Thousand US Dollars (USD 40,000) | | | |
| Total | | Three Million US Dollars (USD 3,000,000) | | | |



Appendix F

Commencement Date APC Partners

Commencement Date APC Partners consist of the following partners.

| Partners | | | |
|--|---|--|--|
| Name | Address | | |
| Emirates Red Crescent | Abu Dhabi Zayed Sports City, PO Box: 633 111 Abu Dhabi-United Arab Emirates | | |
| United Nations Children's Fund (UNICEF) | UNICEF GULF AREA OFFICE, P.O. Box 71456, Dubai, United Arab Emirates | | |